T2R2東京工業大学リサーチリポジトリ Tokyo Tech Research Repository

論文 / 著書情報 Article / Book Information

題目(和文)	高度情報通信ネットワーク基盤の実現に向けた半導体モノリシックマ イクロ波集積回路に関する研究
Title(English)	Research on semiconductor monolithic microwave integrated circuits for the advanced information and communication network infrastructure
著者(和文)	神岡純
Author(English)	Jun Kamioka
出典(和文)	学位:博士(工学), 学位授与機関:東京工業大学, 報告番号:甲第12385号, 授与年月日:2023年3月26日, 学位の種別:課程博士, 審査員:岡田 健一,小寺 哲夫,波多野 睦子,若林 整,廣川 二郎,鈴木 左文
Citation(English)	Degree:Doctor (Engineering), Conferring organization: Tokyo Institute of Technology, Report number:甲第12385号, Conferred date:2023/3/26, Degree Type:Course doctor, Examiner:,,,,,
 学位種別(和文)	博士論文
Type(English)	Doctoral Thesis

Research on semiconductor monolithic microwave integrated circuits for the advanced information and communication network infrastructure

Jun Kamioka

A dissertation submitted for the degree of Doctor of Engineering

Supervised by

Associate Professor Tetsuo Kodera



Tokyo Institute of Technology

March 2023

Abstract

Recently, there has been a need to address social issues via systems that combine communication and sensing in virtual space and reality, using artificial intelligence and big data. In addition, in an age when everything is connected to the network, cyber-attacks are a great threat, and highly reliable and secure communication technologies are becoming even more important. These requirements call for systems that combine ever more advanced wireless communication technologies with new quantum communication technologies as advanced information and communication network infrastructures. Various RF circuits are used in these systems and are keys to achieving system performance. Monolithic microwave integrated circuits (MMICs), where active microwave elements such as field-effect transistors (FETs) and diodes are integrated with passive elements on a single semiconductor chip, can allow for the optimization of RF circuits to meet the miniaturization and performance requirements in these advanced infrastructures. It is crucial to optimize the circuit constants of MMICs and improve the performance trade-off relationship by proposing new circuit configurations and device structures and combining different types of circuits. In this thesis, research is conducted on semiconductor MMICs as a viable route to high-speed, high-capacity, low-power consumption, low-cost, and high-reliability communications.

Key RF devices that would lay the foundation of advanced wireless communication systems combined with emerging quantum communication technologies include highpower and high-efficiency amplifiers and switches, low-cost transmit/receive (T/R) modules, and silicon-based quantum communication and computing chips. The base station uses high-power and high-efficiency GaN amplifiers and switches for wireless communications. GaN devices are key components that account for a large portion of wireless communication systems' power consumption and cost. Electron spin qubits based on silicon quantum dots with long coherence time and the advantage of miniaturization are great candidates for high-reliability quantum networks and computation. Improving the performance of these devices by applying MMIC technologies and optimizing circuit configurations is important, and we have researched as follows.

First, an ultra-wideband bandpass distributed GaN-MMIC power amplifier is designed and measured to realize fast and high-capacity communications. Non-uniform distributed amplifiers (NDPAs) are used as broadband high-power-amplifiers (HPAs), but there has been a trade-off relation between output power and bandwidth. Here, we have improved the trade-off relation by proposing a bandpass distributed amplifier. We designed and evaluated a single-ended bandpass NDPA and a 2-way combined NDPA. Measurement results of the NDPAs show that the developed amplifiers achieve the highest output power and power density among the reported MMIC HPA with relative bandwidth of over 100%.

Second, the design and measurement of a GaN-MMIC high-efficiency power amplifier using the individual source via (ISV) structure is conducted to achieve low power consumption. FETs used in HPAs have a comb electrode structure with multiple source fingers, drain fingers, and gate fingers to increase the total gate width. In conventional FETs, the source fingers are each bundled and connected to the via holes through transmission lines. In this configuration, the parasitic resistance and inductance between the source fingers and the via holes degrades the amplifiers' gain and efficiency. Therefore, an ISV structure with small via holes in all source fingers was applied to improve the performance of GaN amplifiers. We also optimized an output matching circuit by considering the relationship between load impedance and circuit loss. Measurement results of the developed X-band GaN-MMIC amplifier demonstrate the highest performance in terms of the combinations of output power and power added efficiency (PAE) compared to existing state-of-the-art X-band MMIC HPAs.

Third, power amplifiers and a switch incorporating partial MMIC and GaN-on-Si technologies are developed. GaN power amplifier generally refers to GaN fabricated on a SiC substrate (GaN-on-SiC). It has high performance, but tends to be expensive because of the high cost of SiC substrates. Therefore, we developed a GaN-on-Si-MMIC using a cost-effective silicon substrate. We also proposed a partial MMIC configuration that combines a GaN-based MMIC and low-cost, low-loss GaAs matching circuits. These MMIC configurations were combined into a chipset for low-cost T/R modules: a GaN-on-Si-MMIC DA, a GaN-on-SiC MMIC HPA with a GaAs MMIC input and output matching circuits, a high-gain GaN-on-Si HPA with a GaAs output matching circuit and a GaN-on-Si high-power switch. The T/R module configuration using these components achieves a comparable performance to conventional modules at about half the cost.

Finally, research on silicon quantum dot devices for realizing quantum computers and networks is conducted. Design of compact and wideband on-chip matching circuits of quantum dot charge sensor based on RF reflectometry towards fast readout of qubit state is implemented. The matching circuit of a charge sensor has so far been comprised of a shunt capacitor and a series inductor on the printed circuit board using chip components. To reduce the size and broaden the bandwidth of the matching circuit, we extracted the equivalent circuit parameters on the silicon chip of the quantum dots device. We also formulated circuit constants of three matching circuits, the shunt capacitor-series inductor, shunt inductor-series capacitor and shunt inductor-series inductor. We compared them in terms of sensitivity and frequency bandwidth. The newly proposed shunt inductor-series inductor type was shown to have the widest bandwidth. The design with this configuration provides the bandwidth required for high-speed readout while using smaller inductance than conventional circuit configurations, demonstrating the feasibility of on-chip compact matching circuits.

The proposal of new MMIC configurations and analysis in these researches have achieved great performance among the other reported results. We believe that these results will contribute to the realization of the advanced information and communication network infrastructure.

Acknowledgements

As the days of my Ph.D., there are many people I must thank for helping me achieve the results described here.

First, I thank Associate Prof. Tetsuo Kodera for supervising my Ph.D. thesis. He has been a great help to me regarding my research since I was an undergraduate student. Six years after graduating with a master's degree, Kodera Laboratory and Mitsubishi Electric Corporation, where I work, have started a joint research project. A year later, I was accepted as a working doctoral student and received great help in my thesis.

My deepest appreciations go to Prof. Kenichi Okada, Prof. Mutsuko Hatano, Prof. Hitoshi Wakabayashi, Prof. Jiro Hirokawa, and Associate Prof. Safumi Suzuki for examining this thesis.

I am indebted to Project Associate Prof. Jun Yoneda for his insightful and inspiring comments and suggestions.

I want to thank Dr. Raisei Mizokuchi. We have been classmates since our undergraduate days and remained close after graduation. I am happy to be able to do research together after a long time as working doctoral students. I am very grateful for all the support I received in the laboratory.

I would also like to thank Mr. Ryo Matsuda, who worked with me on circuit design and experiments.

I thank my supervisors at Mitsubishi Electric Corporation, Dr. Koji Yamanaka, Dr. Shintaro Shinjo, and Dr. Masaomi Tsuru. They gave me valuable advice and discussions on microwave circuits technology and opportunities for joint research with Kodera laboratory during my Ph.D.

I am also very much indebted to my colleagues at the Mitsubishi Electric Corporation,

Dr. Masatake Hangai, Dr. Yukinobu Tarui, Mr. Yoshitaka Kamo, Mr. Shinichi Miwa, Dr. Yoshifumi Kawamura, Mr. Ryota Komaru, Dr. Kazutomi Mori, and Dr. Yuji Komatsuzaki for discussions and support regarding the design and evaluation of various microwave circuits.

I also thank the other ex- and present lab members. I am deeply grateful to Shimpei Nishiyama, Kaito Tomari, Shunsuke Ohta, Ryutaro Matsuoka, Masaru Hirayama, Sayyid Irsyadul Ibad, Wen Chutian, Takumi Yoshikawa, Kazuma Nakagoe, Masato Machida, Tokio Futaya, Go Sakamoto, Yuto Arakawa, and Chihiro Kondo for fruitful discussions.

Last but not least, I thank my family for supporting me throughout the years.

Table of contents

Chapter 1. Introduction
1.1 Research background1
1.2 Key components for the advanced information and communication network
infrastructure
1.2.1 Wireless transmitting microwave circuits 4
1.2.2 Power amplifiers
1.2.3 Quantum computers and quantum network systems
1.2.4 Semiconductor quantum dot9
1.3 Organization of this thesis11
Chapter 2. Challenges for advanced information and communication network
infrastructure
2.1 Monolithic microwave integrated circuits
2.2 Power amplifier
2.2.1 Basics of power amplifier14
2.2.2 Wideband amplifier 15
2.2.3 High-efficiency amplifier
2.2.4 Low-cost amplifier
2.3 Silicon quantum dot
2.3.1 Charge state in a quantum dot
2.3.2 Charge sensing
2.3.3 RF reflectometry
Chapter 3. GaN-MMIC ultra-wideband bandpass distributed amplifier

3.1 Intro	luction	
3.2 Theo	retical analysis of the artificial line of the distributed amplifier.	
3.3 Desig	n and measurement results of single-ended GaN-MMIC HPA	
3.4 Conc	lusion	
Chapter 4.	GaN-MMIC high-efficiency high power amplifier using indivi-	dual source
	via structure	44
4.1 Intro	luction	44
4.2 Indiv	idual source via structure	45
4.3 GaN	HEMT characteristics	47
4.4 Desig	gn of GaN-MMIC HPA	49
4.5 Meas	urement results	53
4.6 Conc	lusion	56
Chapter 5.	Power amplifiers and a switch incorporating partial MMIC and	nd GaN-on-
	Si technologies for low-cost transmit/receive module	57
5.1 Intro	luction	57
5.1 Intro 5.2 Conc	luction epts of GaN Chipsets	
5.1 Intro 5.2 Conc 5.3 GaN-	luction epts of GaN Chipsets on-Si-MMIC DA	57 58 61
5.1 Intro 5.2 Conc 5.3 GaN- 5.3.1	luction epts of GaN Chipsets on-Si-MMIC DA Circuit Design	57 58 61 61
5.1 Intro 5.2 Conc 5.3 GaN- 5.3.1 5.3.2	luction epts of GaN Chipsets on-Si-MMIC DA Circuit Design	57 58 61 61 62
5.1 Intro 5.2 Conc 5.3 GaN- 5.3.1 5.3.2 5.4 GaN-	luction epts of GaN Chipsets on-Si-MMIC DA Circuit Design Measurement on-SiC HPA with GaAs input and output matching circuits	
5.1 Intro 5.2 Conc 5.3 GaN- 5.3.1 5.3.2 5.4 GaN- 5.4.1	luction epts of GaN Chipsets on-Si-MMIC DA Circuit Design Measurement on-SiC HPA with GaAs input and output matching circuits Circuit Design	
5.1 Intro 5.2 Conc 5.3 GaN- 5.3.1 5.3.2 5.4 GaN- 5.4.1 5.4.2	luction epts of GaN Chipsets on-Si-MMIC DA Circuit Design Measurement on-SiC HPA with GaAs input and output matching circuits Circuit Design	
5.1 Intro 5.2 Conc 5.3 GaN- 5.3.1 5.3.2 5.4 GaN- 5.4.1 5.4.2 5.5 GaN-	luction epts of GaN Chipsets on-Si-MMIC DA Circuit Design Measurement on-SiC HPA with GaAs input and output matching circuits Circuit Design Measurement on-Si HPA with GaAs output matching circuits	
5.1 Intro 5.2 Conc 5.3 GaN- 5.3.1 5.3.2 5.4 GaN- 5.4.1 5.4.2 5.5 GaN- 5.5.1	luction epts of GaN Chipsets on-Si-MMIC DA Circuit Design Measurement on-SiC HPA with GaAs input and output matching circuits Circuit Design Measurement on-Si HPA with GaAs output matching circuits Circuit design	
5.1 Intro 5.2 Conc 5.3 GaN- 5.3.1 5.3.2 5.4 GaN- 5.4.1 5.4.2 5.5 GaN- 5.5.1 5.5.2	luction epts of GaN Chipsets on-Si-MMIC DA Circuit Design Measurement on-SiC HPA with GaAs input and output matching circuits Circuit Design Measurement on-Si HPA with GaAs output matching circuits Circuit design	

5.6.1	Circuit design	72
5.6.2	Measurements	76
5.7 Evalua	ation of cost-effectiveness	77
5.8 Concl	usion	78
Chapter 6.	On-chip matching circuits for RF reflectometry using Si quantum	ı dot
	charge sensor	79
6.1 RF ret	flectometry	79
6.2 Devic	e	81
6.3 Measu	irements	82
6.4 Circui	it design and analysis	86
6.5 Concl	usion	93
Chapter 7.	Conclusion	94

Chapter 1.

Introduction

1.1 Research background

Communication is reproducing at one point either exactly or approximately a message selected at another point [1]. The Morse telegraph was the first to use electric signals for ultra-high-speed communication in 1893. Since then, wireless communication, which enables communication between physically distant equipment such as mobile devices or satellites, and ultrahigh-speed, high-capacity optical communication, have been combined to realize systems that connect globally. New applications and services have been created among such systems by increasing the capacity and speed of communications. Communication is a core technology that changes the face of people's lives. There has recently been need to solve social issues through systems that combine communication and sensing in virtual space and reality, using artificial intelligence and big data. In addition, in an age when everything is connected to the network, cyber-attacks are a great threat, and highly reliable and secure communication technologies are becoming even more important.

Cyber-physical systems (CPS) have been proposed as a new generation of systems with integrated computational and physical capabilities that can interact with humans through many modals [2]. Fig. 1-1 shows the targeted cyber-physical systems. In the physical system, many devices are expected to be connected to the network and optimally controlled. Driverless cars [3] and unmanned aerial vehicles (UAVs) [4] will be moving around the city as next-generation transportation. In the industrial sector, there will be a

need for smart factories where automated guided vehicles (AGVs) and autonomous mobile robots (AMRs) work [5], smart farming [6], monitoring and control of infrastructures, telemedicine systems and so on. Components utilized in the system should not be operated individually, but rather the physical and behavioral interactions between components should be considered to achieve efficient and safe operation of the system as a whole.

To realize CPS, wireless communication infrastructure that can manage acceptable dynamic traffic load and deliver real-time data is necessary. Reliability and robustness are also very important, especially for monitoring and controlling systems because data transmission interruption will cause the unstable operation and cascade failure [8]. There are also challenges in analysis and simulation for overall system optimization in cyber systems. Multiple information must be handled for many devices, resulting in a complex system with enormous variables. Quantum computing, which utilizes quantum superposition is expected to play an active role in such combinatorial optimization with huge amounts of data and dimensions [10][11]. CPS monitor and control various physical processes in major industrial and critical infrastructure, including power grids, energy generating stations, gas pipelines, and water networks. Security risks to critical infrastructure are increasing with the realization of CPS [12]. Therefore, a highly reliable quantum network that can retain security even against attacks using quantum computers is needed.

These requirements call for systems that combine ever more advanced wireless communication technologies with new quantum communication technologies as advanced information and communication network infrastructures. Various RF circuits are crucial to achieving system performance in these systems. Monolithic microwave integrated circuits (MMICs), where active microwave elements such as field-effect transistors (FETs) and diodes are integrated with passive elements on a single semiconductor chip, can optimize the RF circuits as a whole to meet the stringent miniaturization and performance requirements in these advanced infrastructures. The various characteristics of MMICs (frequency bandwidth, output power, efficiency, cost, etc.) have a trade-off relationship, and the method of optimizing circuit constants in a typical circuit configuration has limited performance improvement. Here, we proposed a new circuit configuration, improved the device structure, or combined multiple circuits to improve the trade-off relationship.



Fig. 1-1. Targeted cyber-physical systems

1.2 Key components for the advanced information and communication network infrastructure

Research on the following components was implemented to achieve the advanced information and communication network infrastructures that combine ever more advanced wireless communication technologies with new quantum communication technologies. (1) GaN-MMIC ultra-wideband bandpass distributed amplifier, (2) GaN-MMIC high-efficiency high power amplifier (HPA) using individual source via structure (ISV), (3) Power amplifiers and a switch incorporating partial MMIC and GaN-on-Si

technologies for low-cost transmit/receive module, (4) On-chip matching circuits for RF reflectometry using Si quantum dot charge sensor. Their background and requirements are discussed.

1.2.1 Wireless transmitting microwave circuits

The market for wireless communications has experienced tremendous growth recently. Transmission speeds have increased tenfold approximately every five years to accommodate mobile traffic growth. Next-generation communication systems use multicarrier modulation, such as orthogonal frequency division multiplexing (OFDM) to achieve higher-speed communication. The number of subcarriers is increasing, and the requirement for operating frequency bandwidth is getting wider. Although it is necessary to increase the number of base stations to achieve communication with more devices, simply increasing the number of base station facilities makes installation and operating costs expensive. Therefore, co-construction and shared base station is proposed [14]. The concept of a virtual base station [15] is also important technology. Conventionally, base station facilities composed of a central unit (CU), distributed unit (DU) and radio unit (RU) have been developed for each frequency band and each telecommunications carrier. A virtualized base station is a technology to operate base station equipment suitable for each carrier by switching software on common hardware. While the virtualization of CU and DU has already been developed, RUs have not been virtualized owing to the difficulty of developing several RF components, such as a wideband amplifier. Ideally, a virtualized RU that can accomplish multiple frequency RUs with one is required. In about ten years, fully virtualized base stations will be required to further reduce installation and operation costs. Furthermore, virtualized RUs can switch operational frequencies by software to avoid interference with neighboring base stations.

There are various components to be developed for the realization of virtualized RU.

Tunable filters are needed to suppress unwanted emissions from RF-ICs, antennas, and stray signals from transmit module to receive module. Various configurations of filters have been considered [16] to improve the trade-off relationship among bandwidth, tunability and loss. Reduction of loss of the filter used for the output side of a high-power amplifier and the input side of a low-noise amplifier is particularly important because of its significant impact on output power, efficiency, and noise figure. Hence, research on the tuner filter for the virtualized RU on the filter is still challenging. Wideband high power switches for switching between 4G and 5G systems, transmit/receive mode are necessary. Although loss reduction is still challenging for broadband switches, the bandwidth required for RU virtualization has already been developed [17]. Wideband antennas that can handle various frequency bands in one are also important. Remarkable results have been achieved for wideband antenna technology [18]. Wideband low noise amplifiers (LNAs) and HPAs for received and transmitted signals are also needed. The technology for wideband LNAs is relatively mature and has already been developed for the required bandwidth [19]. For HPAs, it is challenging to obtain wideband and high power for the realization of virtualized RUs. Realizing wideband HPA is one of the most significant challenges for RU virtualization. Reducing power consumption is also important. There are several ways to reduce the base station's power consumption, such as lowering the digital components' operation voltage, digital pre-distortion [20], and PAPR reduction [21]. It is reported that cellular phone base stations consume more than 50% of the energy in a cellular network [22], and 65% of the energy consumed by base stations comes from power amplifiers [23]. Therefore, improving the efficiency of the power amplifiers is the key issue. The number of base stations has increased two to three times in the past decade. Because many T/R modules are installed in active phased allay antennas (APAAs) used for satellite communications or 5th generation mobile network (5G) base stations, reducing the cost per module is also important.

1.2.2 Power amplifiers

Many semiconductor microwave circuits are used in wireless communication systems. Among them, power amplifiers require selecting an appropriate semiconductor process depending on the required performance and application. Power amplifiers are components with high power consumption, which affects the design of the module's heat dissipation. Reliability to withstand high voltage, large RF power, and high channel temperatures is necessary. Table 1-1 shows semiconductor material parameters used in power amplifiers [24][23]. The larger the band gap, the better the chemical stability and the higher the breakdown voltage. GaN is called wide bandgap semiconductors, which allow operation at higher voltages, resulting in higher power density and superior devices for high-power applications. The higher the mobility and electron saturation velocity, the better the high-frequency properties. Fig. 1-2 shows trends in amplifier output power and frequency [25]. This graph shows the survey results from 2000 to November 2020, and the trend toward even higher frequencies and output power continues to grow. There are advantages and disadvantages of using each material in amplifiers, and the most suitable device should be used for each application based on trade-offs. For high-power applications, Si laterally diffused metal oxide semiconductor field effect transistors (LDMOS) are very competitive devices in applications such as mobile phone base stations mainly up to the S-band, due to their low cost and the benefit of highly developed process technology for LSIs. GaAs pseudomorphic high electron mobility transistors (PHEMTs) have been used in HPAs over approximately 100 GHz because of their high mobility, about five times greater than Si. PHEMTs have been used in high-power amplifiers; GaAs HEMT also have excellent noise characteristics and are often used as low-noise amplifiers. Recently, GaN HEMTs have attracted much attention owing to the demand for even higher efficiency and miniaturization of amplifier devices. GaN has a higher operating voltage than GaAs, leading to a higher power density. GaN HPA for base

stations, communication satellites, radar, etc. In recent years, amplifiers with even higher output power, efficiency, and wider bandwidth have been required for advanced communication systems. As obtaining high performance of the GaN HEMT themselves, advanced GaN-MMIC design techniques that maximize the characteristics of the devices are also important.

		Semiconductor			
Characteristic	Unit	Si	GaAs	InP	GaN
Band gap	eV	1.1	1.42	1.35	3.49
Electron mobility	cm ² /Vs	1500	8500	5400	2000
Saturated electron velocity	10 ⁷ cm/s	1	1.3	1	2.5
Critical breakdown field	MV/cm	0.3	0.4	0.5	3.3
Thermal conductivity	Wcm/K	1.5	0.5	0.7	1.5
Relative dielectric constant		11.9	12.8	12.5	9

Table 1-1. Semiconductor material parameter



Fig. 1-2. Trends in amplifier output power and frequency [25]

1.2.3 Quantum computers and quantum network systems

In advanced information and communication network infrastructures, quantum computers that enable ultra-fast computation should be combined with highly reliable quantum communications that are not vulnerable to sophisticated cyber-attacks. A quantum computer is a computer based on a new principle that uses the properties of quantum mechanical superposition and quantum entanglement to solve problems such as prime factorization [26] and database searching [27], which cannot be solved in realistic time using classical computers. In classical computers, the binary values "0" and "1" are used as bits for computation. A superposition of quantum states $|\Psi\rangle = \alpha |0\rangle + \beta |1\rangle$, called a qubit is adopted in quantum computers, where α and β represent complex constants while $|0\rangle$ and $|1\rangle$ are quantum states. Quantum computing on a small scale has been achieved so far [28]-[31]. Quantum error correction is necessary to maintain quantum coherence. However, more than 1000 logic qubits are needed to perform tasks that outperform classical computing, and 10^4 per logic qubit (with an error rate of 1%) is said to be needed to provide quantum error correction (QEC) [32]. Technical challenges remain for quantum computing, such as realizing qubits at high yields, efficient gate manipulation of many qubits, and qubit integration.

Quantum communication is one of the most important technologies for advanced network infrastructures. Various confidential information is connected to networks and shared. Based on the difficulty of the prime factorization, the RSA cipher still plays a central role in the cryptographic infrastructure. The RSA cipher can be solved in polynomial time by using a quantum computer. Quantum cryptography has been proposed as a robust cryptosystem even when large-scale quantum computers become practicable [34]. In wired optical networks, the signal is reduced by losses of cables. In classical communication, optical amplifiers or optical repeaters are used to solve the problem. However, in quantum communication, quantum states are quite fragile, and the quantum no-cloning theorem makes it difficult to amplify or repeat the signal. For a future highperformance quantum internet, quantum error correction will be performed at each relay. As quantum error correction requires many quantum bits, the network will comprise quantum computers connected at each node [33]. By realizing the quantum network, distributed quantum computation for more complex and advanced computations will be achieved. Cloud processing to preserve privacy can also be possible, which is one solution to the problem of communication security.

1.2.4 Semiconductor quantum dot

There has been research on various types of qubits, each system has advantages and disadvantages, and the various qubit systems are expected to be incorporated into suitable fields and combined as a quantum network system.

Superconducting qubit is one of the most promising qubits, with remarkable progress in achieving 53 qubit operations [28]. Superconducting qubit has a resonant circuit structure with a capacitor and a non-linear inductor using a Josephson junction. They have long coherent times, good manufacturability, and the capability of gate manipulation of qubits. The challenge for them is integration to achieve a large-scale qubit system because the footprint of one qubit is relatively larger than other qubit systems (about 1 mm²). The system must be operated at cryogenic temperatures of several 10 mK. It is challenging to realize millions of qubits towards quantum error correction, because of the limited size of the cryogenic space.

Trapped ion qubits are also good candidates for quantum computation. The qubits use charged ion held in an electromagnetic field. They have long coherence times and more than ten qubits have been realized [29][30]. The system also has a scalability problem because controlling many ions in a single ion trap is difficult, and techniques to connect multiple ion traps are required.

Photon qubits have a long coherence time and operation capability at room temperature. They can be transmitted over long distances and are useful for forming quantum networks. However, they are difficult to integrate, and gate operations for multiple qubits cannot be performed efficiently. Research on converting photon qubits to electron spin qubits is also conducted.

Electron spin qubits have long coherence times and compact sizes (10–100 nm). NV centers are one of the promising devices for quantum sensors and quantum repeaters, and qubits based on electron spins confined in quantum dots are also being investigated. NV centers have a long spin coherence time and can operate at room temperature, making them promising devices for quantum sensors and quantum repeaters.

Semiconductor quantum dots (QDs) are promising as quantum computers. QDs can be formed at targeted locations using semiconductor microfabrication processes, which have been developed over the years. Their size is minimal, and their potential and coupling between them can be controlled by gate manipulation. Thus, semiconductor QD can solve the problems of high integration and controllability that have been challenges for other qubits.

Semiconductor quantum dots (QDs) were first intensively studied using GaAs/AlGaAs heterostructures. However, spins in GaAs QD have short coherence time due to hyperfine interaction with nuclear spins, making high-fidelity operation difficult. Therefore, research on QDs using silicon, a material with fewer nuclear spins, has been actively conducted recently. In addition, silicon has a larger effective mass than GaAs, which requires smaller quantum dot sizes. Although the process is more challenging, long coherence times have been confirmed using silicon quantum dots [35][36].

For the readout of spin qubits, charge sensing plays an important role. Charge sensing should be performed within the spin operation time not to limit quantum computation speed. Therefore, RF reflectometry is used to achieve a fast readout of qubits [37]. This

method has the advantage of being insensitive to low-frequency noise, such as 1/*f* noise [38]. Broadband and miniaturization of impedance matching circuits for RF reflectometry is important for high-speed readout and high integration.

1.3 Organization of this thesis

Fig. 1-3 shows the organization of this thesis. Chapter 1 is the background and introduction to the research. Chapter 2 shows the challenges of advanced information and communication network infrastructure. Chapter 3 shows the GaN-MMIC ultra-wideband bandpass distributed amplifier for ultra-fast, large capacity wireless communication systems. Chapter 4 shows the GaN-MMIC high-efficiency high power amplifier using ISV structure to lower power consumption of wireless communication systems. Chapter 5 shows power amplifiers and a switch incorporating partial MMIC and GaN-on-Si technologies for low-cost T/R module. Chapter 6 shows the on-chip matching circuits for RF reflectometry using Si quantum dot charge sensor for high reliability quantum network. Chapter 7 concludes the study.



Fig. 1-3. Organization of this thesis

Chapter 2.

Challenges for advanced information and communication network infrastructure

2.1 Monolithic microwave integrated circuits

MMICs are one of the technologies that enable miniaturization and higher performance of RF circuits. MMIC integrates active microwave elements like FETs and diodes with passive elements such as matching circuits on a semiconductor chip. In contrast, the method in which only active components are formed on a chip and other passive components are formed on a separate printed circuit board (PCB) is called "discrete". In MMIC, compact circuit components such as spiral inductors and metal-insulator-metal (MIM) capacitors can be used. This allows greater freedom in designing RF circuits and enables wider bandwidth and higher efficiency, as there is no need to consider parasitic components such as bonding wires for connection to the PCB. MMIC amplifiers were initially mainly GaAs and SiGe MMICs, but recently, GaN-MMICs have often been developed due to the demand for higher performance. Matching circuits of RF reflectometry circuits in Si quantum dot devices are currently realized in discrete form, but MMICs are expected to achieve miniaturization and higher performance. MMIC design technology is, therefore, very important.

2.2 Power amplifier

2.2.1 Basics of power amplifier

Fig. 2-1 shows a schematic of a simple amplifier. The amplifier comprises a transistor, an input matching circuit, and an output matching circuit. The RF signal is input to the gate of the FET through the input matching circuit, and the amplified signal is output from the drain of the FET through the output matching circuit. The input matching circuit transforms the source impedance (Z_0) to impedance Z_{in} to maximize the gain. The output matching circuit transforms the load impedance (Z_0) to impedance Z_{out} to maximize the output power or efficiency. Owing to the parasitic capacitance in the FET, the optimum load impedance is frequency dependent, and designing a broadband matching circuit is difficult. The optimum load impedance for the transistor is determined by load-pull measurement [39], which performs large-signal measurements while changing input and output loads using an impedance tuner. The basic properties of an amplifier include frequency bandwidth, output power P_{out} , Gain ($P_{out} - P_{in}$), power consumption P_{dc} , and drain efficiency which indicates how much output power is obtained relative to power consumption shown as

$$DE = \frac{P_{out}}{P_{dc}} \times 100$$
 (2-1)

and power-added efficiency, including input power

$$PAE = \frac{P_{out} - P_{in}}{P_{dc}} \times 100.$$
 (2-2)



Fig. 2-1. Schematic of simplified amplifier.

2.2.2 Wideband amplifier

Sub-6 (below 6 GHz) and millimeter wave bands (28 GHz) are used for IoT communications. Millimeter wave is suitable for high-speed communication, but its high linearity makes it difficult for radio waves to penetrate. It is also difficult to achieve high output power and cover a wide area. This study attempts to achieve wideband and high amplifier power for RU virtualization at sub-6 GHz. Until now, different base stations have been used for different generation mobile networks, such as 4G and 5G, with a bandwidth of up to 200 MHz. As explained in 1.2.1, wideband RU is required for the fully virtualized base stations.

In the sub-6 band, 800–4800 MHz is used for the downlink from 4G and 5G base stations to mobile devices, which is 143% of the relative bandwidth given by the following equation

$$Relative \ bandwidth = \frac{Bandwidth}{Center \ frequency} \times 100.$$
(2-3)

Considering the replacement of conventional base stations, a 40-W class high-power amplifier is necessary, but a 40-W class amplifier that exceeds the relative bandwidth of 100% has not been realized. Therefore, wideband HPAs have become a bottleneck in system development.

Several broadband amplifier configurations are described below. Resistively matched

amplifiers use resistors for impedance matching. For example, Fig. 2-2 shows a resistively matched amplifier with shunt resistors on the input and output sides of an FET. Because resistors do not have frequency dependence, wideband flat characteristics can be achieved [40]. However, the amplifier's efficiency, output power, and noise figure are degraded because of losses caused by the resistor. Therefore, it is used as a multi-stage amplifier when only gain and frequency bandwidth are required.



Fig. 2-2. Schematic of resistively matched amplifier

Negative feedback amplifiers are also known as broadband amplifiers. Fig. 2-3 shows a parallel feedback type negative feedback amplifier with a series RC circuit connected between the gate and drain of an FET. Flat gain and reflection characteristics can be obtained over a wide bandwidth. Particularly in LNA, achieving both low input reflection characteristic and NF matching is possible. Therefore, the structure is combined with a reactively matched amplifier configuration, which will be described later. Negative feedback amplifiers with a decade bandwidth have been realized with a few Watts of output power and tend to have low efficiency [19][41].



Fig. 2-3. Schematic of negative feedback amplifier

Reactively matched power amplifiers (RMPAs) are the most commonly used configuration for high-power amplifiers. Fig. 2-4 shows the schematic of RMPA. In the RMPA, reactive components such as inductors, capacitors, and transmission lines are used to obtain impedance matching between the input or output port and the optimum impedance of the transistors. Multiple FETs are arranged in parallel, and the matching circuits are designed to distribute and combine them symmetrically in a tournament-like manner such that all transistors operate in phase. RMPAs have high power and high efficiency owing to the low-loss matching circuit. Higher output power can be achieved simply by increasing the total gate width of the FETs, and output power above 50 W can be achieved with a GaN-MMIC. The output impedance of an amplifier can be expressed by a parallel circuit of resistors (R_p) and capacitors (C_p) [42]. According to Bode–Fano theory, reactive matching for parallel RC circuits has limited bandwidth [43]. The bandwidth limit of an output matching circuit is approximately expressed in inverse proportion to the supply voltage shown in the following equation [44].

$$BW < \frac{4.343I_{\text{MAX}}}{RL(dB)C_{\text{p}}\beta(V_{\text{D}} - V_{\text{k}})}$$
(2-4)

where V_D is drain voltage; V_K is knee voltage (minimum drain voltage where maximum drain current is obtained); I_{MAX} is maximum drain current; *RL* is return loss, and β is fitting parameters which typically vary between 1 and 2. This is a theoretical limit, and the bandwidth becomes even narrower when a realizable circuit size is considered. Hence, a typical matching circuit has up to three stage impedance transformers, and the relative bandwidth of reported GaN-MMIC amplifiers with over 20-W class output power is up to 100 %.



Fig. 2-4. Schematic of RMPA.

Distributed amplifiers are a configuration that can overcome the effects of the Bode– Fano limitation. Reducing C_p is effective for wider bandwidth. Fig. 2-5 shows the schematic of the uniform distributed power amplifier. The FET's output and input parasitic capacitance is absorbed in the "artificial transmission line" with impedances Z_d and Z_g [45]. Low reflection properties are obtained over a wide bandwidth by designing all artificial transmission lines to have the same impedance as the termination resistors (R_g, R_d) . The disadvantages of uniform distributed power amplifiers are low output power and efficiency because the output power is consumed at the terminating resistor on the output side R_d . In addition, impedance matching between optimum load impedance and output port is not achieved.

To solve this problem, a non-uniform distributed power amplifier (NDPA) was developed [46]. Fig. 2-6 shows schematic of the non-uniform distributed power amplifier. In non-uniform distributed amplifiers, the output side termination resistor is eliminated. The impedance of the artificial transmission line on the output side is higher on the left in the figure to obtain the optimum load impedance for each FET. High output and high efficiency for wide bandwidth can be obtained by NDPA. However, the NDPA does not allow for infinitely wide bandwidth. The parasitic capacitance of the FET and transmission lines have a cut-off frequency, which limits their wideband capability. Reducing the unit gate width of each FET and increasing the number of FETs can increase the cut-off frequency, but it increases circuit loss and size, leading to lower output power and efficiency. Therefore, even with non-uniform distributed amplifiers, bandwidth, output power, and efficiency have a trade-off relationship. Achieving both high output power and wide bandwidth is a challenge.



Fig. 2-5. Schematic of the uniform distributed power amplifier.



Fig. 2-6. Schematic of the NDPA.

2.2.3 High-efficiency amplifier

Communication traffic is increasing year by year. Power amplifiers are said to account for over 30% of the power consumption of a communication system [22][23]. Higher efficiency is required for the design of low-loss output matching circuits and high efficiency of the transistors. The output matching circuit is particularly important for improving efficiency in amplifier design because losses in circuits other than the output matching circuit only reduce the gain. In contrast, losses in the output matching circuit reduce the saturation output power. The key point is trade-offs of circuit configurations and optimization of the circuit parameters of the output matching circuit. Using GaN transistors with high mobility and high voltage operation effectively improves the efficiency of FET. Owing to the high output power density, the number of FETs and the circuit size can be reduced compared to FETs with other materials, leading to lower circuit loss.

Improvements in the epitaxial structure and optimization of the GaN transistor structure, including gate electrodes are also being made to increase efficiency. The source inductance of the transistor causes a negative feedback effect on the transistor, which causes reduced gain and efficiency. A transistor structure that reduces such parasitic components is desirable.

2.2.4 Low-cost amplifier

GaN transistors are generally fabricated on SiC substrates. Because SiC wafer is expensive, the cost of a GaN power amplifier is higher than GaAs or Si power amplifiers. GaN power amplifiers were initially developed primarily in discrete configurations. For discrete GaN transistors, GaN was only used as FET chips, and impedance matching circuits are designed using microstrip lines on the high-k substrate or chip LC components on PCBs by connecting with bonding wires. However, the presence of bonding wires and parasitics in chip components made obtaining high-frequency performance and broadband characteristics difficult. To solve these problems, GaN-MMICs have been actively developed in recent years. GaN-MMIC amplifiers have been realized for millimeter-wave and beyond with great performance although the cost is higher. The choice between discrete and MMIC configurations has been based on a trade-off relationship between cost and performance. However, there has been a particularly strong demand for both low cost and high performance, and improvements in devices and circuit configurations are required.

2.3 Silicon quantum dot

2.3.1 Charge state in a quantum dot

A quantum dot (QD) is an artificial system in which electrons (or holes) are confined in zero-dimensional space. The QDs are connected to leads (electron reservoirs) through a tunnel barrier, and electrons or holes are exchanged between the leads and the dots. By connecting the source and drain leads to QD, it is possible to measure the conduction properties of the QDs. As explained in 1.2.4, silicon quantum dots have recently attracted much attention because of their long coherence time. There are gate-defined silicon QD and physically-defined Si QD. A gate-defined QD is electrically formed in 2DEG by applying a voltage to some fine gates. Gate-defined QDs allow fine adjustment of QD potential, but the large number of gates makes integration difficult. Physically-defined QD is formed by etching silicon on an insulator wafer. This structure is advantageous for integration because fewer gates are needed to adjust the quantum dots.

The electrostatic potential in the QD can be modulated relative to the leads by adjusting the voltage applied to the gate electrode electrostatically coupled to the QD. Fig. 2-7 shows the equivalent circuit of a single QD. A tunnel coupling is represented as parallel tunnel resistor and tunnel capacitor and these values vary depending on the potential of the lead and quantum dots. One of the phenomena characteristically observed in such QD systems is called Coulomb oscillation. The conductivity of quantum dots changes periodically with gate voltage and peaks at a specific gate voltage. The phenomenon in which the conductivity is significantly suppressed between current peaks is called the Coulomb blockade. In the following, we discuss the characteristic conduction properties of these quantum dots using the constant interaction model [42]. The constant interaction model is based on the following two assumptions.

1. The Coulomb interaction between one electron in the dot and all other electrons is

given by a constant capacitance C.

2. The energy levels of a single particle can be calculated in terms of discrete energy levels in the absence of interaction between electrons.

The capacitance C in the first assumption is given by $C = C_S + C_D + C_G$ where C_S , C_D , and C_G are the capacitances between the quantum dot and the source, drain, and gate electrodes, respectively. Under these assumptions, the total energy of a QD containing N electrons is

$$U(N) = \frac{(eN - C_{\rm D}V_D - C_{\rm S}V_{\rm S} - C_{\rm G}V_{\rm G})^2}{2C} + \sum_{j=1}^{N} E_n(B)$$
(2-5)

where *e* is a single electron charge, *N* is the number of electrons in the QD; V_i is the voltage applied to electrode i, and i = D, S, G corresponds to drain, source, and gate. ε_j is j-th eigenenergy of a single-particle Schrödinger equation for the QD confinement. The electrochemical potential required for adding N-th electron, $\mu(N)$, is

$$\mu(N) = U(N) - U(N-1) = \left(N - \frac{1}{2}\right)E_C - \frac{E_C}{e}(C_D V_D + C_S V_S + C_G V_G) + E_N \quad (2-6)$$

where $E_C = e^2/C$, called charging energy. The energy spacing between two sequential electrochemical potentials, referred to as addition energy $E_{add}(N)$, is

$$E_{add}(N) = \mu(N+1) - \mu(N) = E_C + \Delta E.$$
 (2-7)

 $E_{add}(N)$ contains the charging energy and the energy spacing $\Delta E = E_{N+1} - E_N$.



Fig. 2-7. Equivalent circuit of a single QD.

The tunneling current of a quantum dot is determined by the relationship between the dot's electrochemical potential $\mu(N)$ of the dot and the electrochemical potentials of the source and drain electrodes (μ_S and μ_D). Applying a bias voltage $V_{SD} = V_D - V_S$ between source and drain opens a bias window with a width of $|eV_{SD}|$, and tunneling current through the QD flows only when the electrochemical potential of the QD is within this window. When $e|V_{DS}| = |\mu_S - \mu_D| \ll E_C, \Delta E_N$, the QD current is only observed when the electrochemical potential $\mu(N)$ is within the bias window ($\mu_S \leq \mu(N) \leq \mu_D$). When the electrochemical potential of the QD current does not flow and the number of electrons in the dot is maintained. This is called the Coulomb blockade.



Fig. 2-8. Coulomb oscillation. (a) Electrochemical potentials of a single QD in the Coulomb blockade. (b) Electrochemical potentials of a single QD at a Coulomb peak.

(c) Drain-source current (I_{ds}) through QD as a function of gate voltage (V_G)

2.3.2 Charge sensing

Manipulation and readout of electron spin qubits in a QD were performed with a few electrons. In such a state, the current flowing throw the QD is very small, making readout difficult. Therefore, the charge sensing technique was utilized. By placing a charge sensor (CS) QD near of the qubit QD, a CS's I–V characteristics change steeply with respect to the change in the number of electrons in qubit QD. Fig. 2-9 shows the schematic of charge sensing. Fig. 2-9 (a) shows the equivalent circuit of QD and CS QD. QD and CS are capacitively coupled. Fig. 2-9 (b) and (c) shows the drain-source current of QD and CS, respectively. Because the I–V characteristics of CS QD vary steeply with respect to the nearby electric field, when the number of electrons in QD changes, the I–V characteristic

of CS shows "kink". Spin states in QD can be read out using spin-to-charge-conversion [48][49].



Fig. 2-9. Schematic of charge sensing. (a) Equivalent circuit of QD and CS. (b) Drain-source current of QD. (c) Drain-source current of CS.

2.3.3 RF reflectometry

A fast readout of the spin states is necessary to achieve the quantum-non-demolition spin readout for fault-tolerant quantum computing. However, charge sensing based on I–V measurement is typically slow (the order of kHz [37]), and the sensitivity is limited by 1/*f* noise [38]. Charge sensing using RF reflectometry has been widely studied in QD systems [50]–[53] to solve the problem. Fast spin readout within coherence times was realized in a gate-defined Si/SiGe QD with RF reflectometry using QD CS [50]. In physically-defined QD with fewer gates than gate-defined QD, it is more difficult to adjust tunneling resistance; hence accurate circuit design is more important.

RF reflectometry reads the change in microwave reflection characteristics of the CS QDs. As mentioned earlier, a parallel RC circuit represents the equivalent circuit of the
quantum dot charge sensor. The resistance of the quantum dots is quite high compared to 50 Ω . Therefore, quantum dot's reflection characteristic is almost open. Therefore, the quantum dot chip is connected to an external PCB and a matching circuit with chip components is used for the impedance transformation from 50 Ω to the input impedance of QD at a bias condition. Charge sensing can be achieved by reading the reflection amplitude or phase change when the resistance or capacitance changes. In conventional reflectometry, impedance matching is performed using a chip series inductance and a shunt capacitor (mainly parasitic capacitance of the PCB). In this configuration, the PCB's parasitic shunt capacitance determines the operation frequency and value of inductance. For a QD impedance of several M Ω , the operation frequency is around 100 MHz, and the size of the inductor used in the matching circuit is the order of μ H, requiring a large chip inductor. For future frequency multiplexing, it is desirable to have a circuit configuration that enables matching at various frequencies. In addition, for fast readout, it is necessary to have a wideband impedance matching. Changes in the reflectance characteristics of quantum dots are read out as pulsed signals. The required bandwidth is proportional to the inverse of the readout time. Considering QEC, a state fidelity of 99% [32] or more is required. The state fidelity of data qubits is expressed by the following equation

$$e^{-\frac{t}{T_2^H}} \cong 1 - \frac{t}{T_2^H}$$
 (2-8)

By assuming the dephasing time T_2^H of 130 µs [54], readout time of less than 1.3 µs is required. To achieve that, a bandwidth of 0.77 MHz or more is required for the matching circuit of the RF reflectometry. Considering QEC, the number of matching circuits for the charge sensor also increases; hence it is necessary to downsize the circuit. The application of MMIC technology, which can be made at a higher frequency will be a good solution.

Chapter 3.

GaN-MMIC ultra-wideband bandpass distributed amplifier

There is a need for high-capacity, high-speed communications and base stations that can be shared across frequency bands for multiple generations of communication systems. To meet these demands, microwave amplifiers with a wide bandwidth and high output power are required. As described in Section 2.2.2, there is a trade-off between bandwidth and output power in microwave amplifiers. In this chapter, we propose a bandpass distributed amplifier that improves the tradeoff relationship. We also design and evaluate a wide-bandwidth, high-power bandpass-distributed amplifier with a specific bandwidth of 120% or more in the S-X band, and confirm the effectiveness of the proposed circuit.

3.1 Introduction

HPAs with high efficiency and wide operation bandwidth are in increasing demand for radar or communication systems. GaN high electron mobility transistors (HEMTs) have been utilized in HPAs because of its high power capability. To date, wideband GaN-MMIC HPAs with various circuit configuration have been developed [55]–[71]. NDPAs were proposed to enhance output power and efficiency. Optimum load impedance for each FET is obtained by properly tapered characteristic impedance of "artificial transmission lines" [71]. Utilizing FET cells with large gate width is effective to increase output power of NDPA but cut-off frequency decrease due to large parasitic capacitor of

FETs. In this chapter, bandpass distributed amplifier configuration is applied to NDPAs in order to increase cut-off frequency without reducing gate width of FETs. Two MMIC HPAs are designed and measured. One is a single-ended HPA which achieves output power of 17 to 26 W, PAE of 24 to 44% across 2.5 to 11.0 GHz. The other is a two-way combined HPA which achieves output power of 27 to 61 W, PAE of 24 to 43% across 2.5 to 10.0 GHz. These results demonstrate the highest output power among wideband amplifiers ever reported.

3.2 Theoretical analysis of the artificial line of the distributed amplifier

The advantage and design method of a bandpass NDPA are described here. Fig. 3-1 shows a schematic of a conventional NDPA. Input and output matching circuits of the NDPA have artificial transmission lines composed of inductance of transmission lines and parasitic capacitance of FETs. For simplicity, design method is explained focused on the input circuit. Impedance of each artificial transmission line in the input circuit is the same impedance as the termination resistance. This realizes broadband characteristic overcoming Bode-Fano criterion [72], [73] which restricts the bandwidth of reactively-matched power amplifiers (RMPAs). Although, the bandwidth of a NDPA is limited by cut-off frequency of the transmission lines due to parasitic capacitance of FETs. To discuss bandwidth, cut-off frequency of the artificial transmission lines are calculated. Fig. 3-2 shows the LC circuit which is simplified equivalent circuit of artificial transmission line of the conventional distributed amplifier.



Fig. 3-1. Schematic of a conventional NDPA.



Fig. 3-2. Simple equivalent circuit of unit artificial transmission line (LC circuit)

F-matrix of the LC circuit is given by

$$\begin{bmatrix} A & B \\ C & D \end{bmatrix}_{Unit} \equiv \begin{bmatrix} 1 & Z \\ 0 & 1 \end{bmatrix} \begin{bmatrix} 1 & 0 \\ Y & 1 \end{bmatrix}$$
$$= \begin{bmatrix} 1 & j\omega L_{\rm R} \\ 0 & 1 \end{bmatrix} \begin{bmatrix} 1 & 0 \\ j\omega C_{\rm R} & 1 \end{bmatrix}$$
$$= \begin{bmatrix} 1 - \omega^2 C_{\rm R} L_{\rm R} & j\omega L_{\rm R} \\ j\omega C_{\rm R} & 1 \end{bmatrix}.$$
(3-1)

When the impedance of input and output ports are Z_0 , S_{21} of the LC circuit is

$$S_{21Unit} = \frac{2}{2 - \omega^2 C_R L_R + j \omega L_R / Z_0 + j \omega C_R Z_0}.$$
 (3-2)

By assuming the input port and output port and the characteristic impedance are the same, their relation is given by

$$Z_{\rm in} = Z_{\rm out} = \sqrt{\frac{L_{\rm R}}{C_{\rm R}}}.$$
(3-3)

By defining

$$\omega_{\rm se} \equiv \frac{1}{\sqrt{C_{\rm R}L_{\rm R}}},\tag{3-4}$$

Equation (3-2) is derived to

$$S_{21Unit} = \frac{1}{1 - \frac{1}{2} \left(\frac{\omega}{\omega_{se}}\right)^2 + j \frac{\omega}{\omega_{se}}}$$
(3-5)

and transmission amplitude is calculated to be

$$|S_{21Unit}| = \frac{1}{\sqrt{\left[1 - \frac{1}{2} \left(\frac{\omega}{\omega_{se}}\right)^2\right]^2 + \left(\frac{\omega}{\omega_{se}}\right)^2}}$$
$$= \frac{1}{\sqrt{1 + \frac{1}{4} \left(\frac{\omega}{\omega_{se}}\right)^4}}.$$
(3-6)

Then, 3 dB cut-off frequency of the LC circuit (ω_{LC}) is calculated as follows.

$$\frac{1}{4} \left(\frac{\omega_{LC}}{\omega_{se}} \right)^4 = 1$$
$$\Rightarrow \omega_{LC} = \sqrt{2} \omega_{se} = \sqrt{\frac{2}{C_R L_R}}$$
(3-7)

It is shown that cut-off frequency of input circuit is determined by gate-source capacitance of FET (C_R) since L_R is fixed by (1). Therefore, reducing the gate width of the FET and increasing the number of FETs are effective to increase cut-off frequency but this also increases loss and then the efficiency decrease. To avoid this, additional capacitor C_{add} is inserted in series with C_R in conventional NDPA[55] as shown in Fig. 3-3. In this case, 3 dB cut-off frequency is derived to

$$\omega_{\text{LC-}} = \sqrt{\frac{2}{\frac{C_{\text{R}}C_{\text{add}}}{C_{\text{R}} + C_{\text{add}}}}} \sum_{\text{R}} = \sqrt{\frac{C_{\text{R}} + C_{\text{add}}}{C_{\text{add}}}} \omega_{\text{LC}}.$$
 (3-8)

Equation (3-8) indicates that cut-off frequency can be increased by additional capacitance. Although, gain of FETs decrease because voltage applied to gate-source capacitance is divided. In other words, conventional NDPAs have trade-off relation between gain and cut-off frequency.



Fig. 3-3. Simple equivalent circuit of unit artificial transmission line (LC circuit with additional capacitor)



Fig. 3-4 Simple equivalent circuit of unit artificial transmission line of bandpass distributed amplifier

To improve the trade-off relation in the conventional NDPA, bandpass NDPA configuration is proposed. Fig. 3-4 shows a simple equivalent circuit of artificial transmission lines of the bandpass NDPA. In this configuration, gain is not reduced since gate-source voltage is not divided. Cut-off frequency of the bandpass NDPA is also calculated. Coefficient α (>0) is defined as relation between the LC circuit parameters (L_R, C_R) and additional parameters in the bandpass circuit (L_L, C_L).

$$L_{L} = \alpha L_{R}, C_{L} = \alpha C_{R}.$$
(3-9)

By defining

$$\omega_{\Gamma} = \frac{1}{\sqrt{C_{\rm R}L_{\rm L}}} = \frac{1}{\sqrt{\alpha L_{\rm R}C_{\rm R}}} = \frac{\omega_{\rm LC}}{\sqrt{2\alpha}},\tag{3-10}$$

F-matrix of the bandpass circuit is given by

$$\begin{bmatrix} A & B \\ C & D \end{bmatrix}_{Unit} \equiv \begin{bmatrix} 1 & Z \\ 0 & 1 \end{bmatrix} \begin{bmatrix} 1 & 0 \\ Y & 1 \end{bmatrix}$$

$$= \begin{bmatrix} 1 - \frac{1}{\alpha} \left(\frac{\omega}{\omega_{\Gamma}} - \frac{\omega_{\Gamma}}{\omega} \right)^2 & j \left[\left(\frac{\omega}{\omega_{\Gamma}} \right)^2 - 1 \right] \left(\frac{\omega_{\Gamma}^2 L_R}{\omega} \right) \\ j \frac{(\omega/\omega_{\Gamma})^2 - 1}{\alpha \omega L_R} & 1 \end{bmatrix} .$$

$$(3-11)$$

By assuming that the input port, the output port and characteristic impedance Z_0 , S_{21} of the bandpass circuit is calculated as

$$S_{21Unit} = \frac{1}{1 - \frac{1}{2\alpha} \left(\frac{\omega}{\omega_{\Gamma}} - \frac{\omega_{\Gamma}}{\omega}\right)^2 + \frac{j}{2} \left(\frac{\omega}{\omega_{\Gamma}} - \frac{\omega_{\Gamma}}{\omega}\right) \left(\frac{\omega_{\Gamma} L_R}{Z_0} + \frac{Z_0}{\alpha \omega_{\Gamma} L_R}\right)}.$$
(3-12)

By assuming the characteristic impedance of bandpass circuit is also Z₀,

$$Z_0 = Z_{\rm in} = Z_{\rm out} = \sqrt{\frac{L_{\rm R}}{C_{\rm R}}} = \sqrt{\frac{L_{\rm L}}{C_{\rm L}}}.$$
 (3-13)

From (2-10) and (2-13),

$$\frac{\omega_{\Gamma}L_{\rm R}}{Z_0} = \frac{1}{\sqrt{L_{\rm R}C_{\rm L}}} \sqrt{\frac{C_{\rm R}}{L_{\rm R}}} L_{\rm R} = \sqrt{\frac{C_{\rm R}}{C_{\rm L}}} = \frac{1}{\sqrt{\alpha}}$$
(3-14)

By substituting (2-14) to (2-12),

$$\frac{1}{1 - \frac{1}{2\alpha} \left(\frac{\omega}{\omega_{\Gamma}} - \frac{\omega_{\Gamma}}{\omega}\right)^{2} + j \frac{1}{\sqrt{\alpha}} \left(\frac{\omega}{\omega_{\Gamma}} - \frac{\omega_{\Gamma}}{\omega}\right)} = \frac{1}{1 - \frac{\chi^{2}}{2\alpha} + j \frac{\chi}{\sqrt{\alpha}}}$$
(3-15)

where

$$\chi \equiv \frac{\omega}{\omega_{\Gamma}} - \frac{\omega_{\Gamma}}{\omega}.$$
 (3-16)

Then transmission amplitude is calculated to be

$$|S_{21Unit}| = \frac{1}{\sqrt{\left(1 - \frac{\chi^2}{2\alpha}\right)^2 + \left(\frac{\chi}{\sqrt{\alpha}}\right)^2}} = \frac{1}{\sqrt{1 + \frac{\chi^4}{4\alpha^2}}}$$
(3-17)

Then, 3 dB cut-off frequency ω_c is calculated as follows.

$$\frac{\chi|_{\omega=\omega_{C}}^{4}}{4\alpha^{2}} = 1 \Leftrightarrow \left(\frac{\omega_{c}}{\omega_{\Gamma}} - \frac{\omega_{\Gamma}}{\omega_{c}}\right)^{2} = 2\alpha$$
(3-18)

From (2-10) and (2-17), when $\omega_c > \omega_{\Gamma}$,

$$\omega_{\rm c}^2 - \sqrt{2\alpha}\omega_{\rm r}\omega_{\rm c} - \omega_{\rm r} = 0$$

$$\Rightarrow \omega_{\rm c} = \frac{\sqrt{2\alpha + 4} + \sqrt{2\alpha}}{2}\omega_{\rm r} = \frac{1}{2}\left(\sqrt{1 + \frac{2}{\alpha}} + 1\right)\omega_{\rm LC} \equiv \omega_{\rm c+} \qquad (3-19)$$

When $\omega_{\rm c} < \omega_{\Gamma}$,

$$\omega_c^2 - \sqrt{2\alpha}\omega_{\Gamma}\omega_c - \omega_{\Gamma} = 0$$

$$\Rightarrow \omega_c = \frac{\sqrt{2\alpha + 4} - \sqrt{2\alpha}}{2}\omega_{\Gamma} = \frac{1}{2}\left(\sqrt{1 + \frac{2}{\alpha}} - 1\right)\omega_{LC} \equiv \omega_{c-}.$$
 (3-20)

(19) indicates that the bandpass circuit has lower cut-off frequency (ω_{c} -) and higher cutoff frequency (ω_{c+}). It is shown that the higher cut-off frequency is higher than that of LC circuit since $\alpha > 0$. From (3-19) and (3-20),

$$\omega_{c+} - \omega_{c-} = \omega_{LC}. \tag{3-21}$$

(3-21) indicates that a total bandwidth of the bandpass circuit and the conventional LC circuit are the same. It is apparent that higher cut-off frequency of the bandpass circuit is higher than that of LC circuit by the bandwidth of the lower-cut off frequency. Fig. 4 shows transmission characteristics of LC circuit and bandpass circuit. In actual design, conventional NDPAs with LC circuit configuration has lower cut-off frequency due to the

limitation of realizable on-chip DC-cuts and choke inductors on MMIC. In addition, lower frequency region is usually stabilized to avoid unwanted oscillation. By applying the bandpass amplifier configuration, the bandwidth of wasted lower frequency region can be shifted to increase higher cut-off frequency.

How to determine parameters of bandpass NDPA by modifying already designed conventional LC circuit is described. By solving (3-20) for α , α is given by

$$\alpha = \frac{2}{\left(\frac{2\omega_{c-}}{\omega_{LC}} + 1\right)^2 - 1} = \frac{1}{2A(A+1)}, A \equiv \frac{\omega_{c-}}{\omega_{LC}}$$
(3-22)

By determining lower cut-off frequency, A and then the design parameter α is obtained from (3-22). When $\omega_{c} \rightarrow 0$ ($A \rightarrow 0$), it is apparent that $\alpha \rightarrow \infty$ where the circuit is equivalent to the conventional LC circuit. This means that (3-22) is applicable to the design of conventional LC circuits. Fig. 3-5 shows the relation between α and A.

This design method can be also applied to the output matching circuit by considering C_R as drain-source capacitance.



Fig. 3-5. Relation between α and A.

3.3 Design and measurement results of single-ended GaN-MMIC HPA

Two MMIC HPAs are designed utilizing 0.25 µm GaN HEMT technology on 100 µm thick silicon carbide (SiC) substrate based on proposed design method. One is a singleended bandpass NDPA. Fig. 3-6 shows a schematic diagram of the single-ended bandpass NDPA. 4 FET cells are utilized. Gate peripheries of the FET1–4 are 1.0 mm, 0.7 mm, 0.6 mm, 0.5 mm respectively. Bandpass artificial transmission lines are applied to the FET1 and the FET4 to make the chip size compact. Another advantage of the bandpass NDPA is that each shunt-inductor can be used as a bias circuit. In case of the conventional NDPA, unnecessary inductance of bias circuit degrades the performance.

The other is a HPA realized by combining two single-ended HPAs in parallel. Fig. 3-7 shows the schematic of the two-way combined NDPA. Combiner is composed of 70.7- Ω transmission line transformers with the electrical length of $\lambda/4$ at the frequency of slightly higher than the center frequency to compensate lower gain at high frequency.



Fig. 3-6. Schematic diagram of a bandpass NDPA.



Fig. 3-7 Schematic of the two-way combined NDPA

Fig. 3-8 shows the fabricated single-ended GaN-MMIC HPA. The chip size is 2.8×1.5 mm². The developed single-ended GaN-MMIC HPA is measured in small-signal and large-signal conditions. Fig. 3-9 shows measured S-parameters of the single-ended GaN-

MMIC HPA. Measurements are performed at a DC drain voltage of 50 V and quiescent drain current of 50 mA. The small-signal gain is greater than 12 dB over 2.5 to 11.0 GHz. Input and output return loss is greater than 6.4 dB and 9.6 dB, respectively.

Large-signal measurement of the single-ended GaN-MMIC HPA is performed under pulsed gate bias conditions with pulse duty-cycle of 10%. Fig. 3-10 shows measured saturated output power (P_{out}) and PAE of the developed single-ended GaN-MMIC HPA as a function of frequency. The output power of 42.2 to 44.1 dBm (17 to 26 W, 20 W on average), the PAE of 24 to 44% (35% on average) at input power of 35.5 dBm over 2.5 to 11.0 GHz are achieved.



Fig. 3-8. Photograph of the developed GaN-MMIC NDPA



Fig. 3-9. Measured S-parameters of the single-ended GaN-MMIC NDPA



Fig. 3-10. Measured output power and PAE of the single-ended GaN-MMIC HPA

The developed two-way combined GaN-MMIC HPA is measured in small-signal and large-signal conditions. Fig. 3-11 shows the fabricated two-way combined GaN-MMIC HPA. The chip size is $3.2 \times 3.0 \text{ mm}^2$. Fig. 3-12 shows measured S-parameters.

Measurements are performed at a DC drain voltage of 50 V and quiescent drain current of 50 mA. The small-signal gain is greater than 11 dB over 2.5 to 10 GHz. Input and output return loss is greater than 4.7 dB and 6.6 dB, respectively.

Large-signal measurement of the two-way combined GaN-MMIC HPA is performed under pulsed gate bias conditions with pulse duty-cycle of 10%. Fig. 3-12 shows measured saturated output power and PAE of the developed two-way combined GaN-MMIC HPA as a function of frequency. Output power of 44.3 to 47.9 dBm (27 to 61 W, 40 W on average), PAE of 24 to 43% (31% on average) at input power of 38 dBm over 2.5 to 10.0 GHz are achieved. Table 3-1 summarizes the performance comparison to other broadband MMIC power amplifiers. It shows that the developed amplifier demonstrates the largest output power and power density. Great performance considering combination of output power and PAE compared to broadband MMIC amplifier ever reported is obtained.



Fig. 3-11. Photograph of the two-way combined GaN-MMIC NDPA



Fig. 3-12. Measured S-parameters of the two-way combined GaN-MMIC HPA.



Fig. 3-13. Measured output power and PAE of the two-way combined GaN-MMIC HPA.

Refs.	Frequency [GHz]	Output power [W]	PAE [%]	Area [mm ²]	Power density [W/mm ²]	Circuits
[55]	1.5 – 17	9 – 15	20-38	15.3	0.58 – 0.98	NDPA
[56]	2-20	10-21	15 – 36	38	0.26 – 0.56	2-way NDPA
[57]	2-20	18-20	18-40	N/A	-	RMPA
[58]	2-18	9.1 - 15.8	18-38	7	1.3 – 2.3	NDPA
[59]	1-8	9 – 13	29 – 48	11.4	0.81 – 1.15	NDPA
[60]	1 – 7	14 – 27	26-44	47.4	0.29 – 0.57	2-way NDPA
[61]	2.5 - 10.5	18-37	19-40	20	0.89 – 1.86	RMPA
[62]	2.0 - 18.0	9.1 - 15.8	18.3 - 38.1	7.0	1.3 - 2.26	NDPA
[63]	2.0 - 20.0	18.0 - 30.8	18.0 - 29.7	25.8	0.70 –1.19	2-way NDPA
This work	2.5 - 11.0	17 – 26	24 – 44	4.2	3.9 - 6.1	Bandpass NDPA
	2.5 - 10.0	27 - 61	24 - 43	9.6	2.8-6.4	Bandpass 2-way NDPA

Table 3-1. Performance comparison to other state-of-the-art broadband MMIC high power amplifiers

3.4 Conclusion

Bandpass NDPA to achieve high power and wide bandwidth is proposed. Two bandpass NDPAs are developed and measured. One is a single-ended HPA which demonstrates output power of 17 to 26 W, PAE of 24 to 44% across 2.5 to 11.0 GHz (relative bandwidth of 126%) with chip size of 4.2 mm². The other is a two-way combined HPA which demonstrates output power of 27 to 61 W, PAE of 24 to 43% across 2.5 to 10.0 GHz

(relative bandwidth of 120%) with chip size of 9.6 mm². These results demonstrate highest output power and power density among wideband amplifiers ever reported. We have realized design technology for wideband, high-power amplifiers with relative bandwidths exceeding 120% which enable base station for high-capacity, high-speed communication, or multipurpose radars. Although the developed amplifier is out of the sub-6 frequency, this technology can be scaled to lower frequencies. The target specific bandwidth of 143% mentioned in section 2.2.2 has not been achieved, and further bandwidth expansion is needed. The maximum PAE of the developed amplifier is 43%. In conventional narrowband amplifiers, the backoff PAE is about 40% [65]. Although the developed amplifier has a lower backoff PAE than 43%, the envelope tracking technique allows the amplifier to maintain its efficiency during backoff. When the efficiency of the power modulator used for envelope tracking is 78% [66], the backoff PAE is estimated to be about 33%. It is basically difficult to achieve the same efficiency in a wideband as a narrowband amplifier, but efforts must be made to get close enough to enable simple replacement. Improving the efficiency is important and will be discussed in Chapter 4. This circuit configuration allows the cutoff frequency to be changed as described in 3.2, and expansion to various frequencies can be realized within the frequency range in which the FET can be operated.

Chapter 4.

GaN-MMIC high-efficiency high power amplifier using individual source via structure

It is said that more than 30% of the power consumption of communication systems is due to power amplifiers [22][23], and high efficiency HPAs is strongly required. GaN transistors are widely used as high-efficiency HPAs in communication base stations, communication satellites, radars and so on. Comb electrode structure with gate fingers between multiple source and drain fingers is typically used for high-power amplifiers. In the conventional FET structure, the gain and efficiency characteristics deteriorate at higher frequencies due to parasitic resistance and inductance in the transmission line from the source finger to the via hole. In this chapter, high efficiency GaN-MMIC HPA using the FET with ISV structures [85] is used to solve this problem. The ISV structure has fine via holes in each source finger section, which has low source resistance and inductance to improve gain and efficiency characteristics.

4.1 Introduction

T/R modules have been widely used in radar, satellite communications, base stations for mobile communication systems, and other applications. To lower the cost and increase versatility, the size of these systems must be reduced. Accordingly, the heat problem is getting more serious. As HPAs are one of the most power-consuming components in T/R modules, there is a high demand for an HPA of a compact size and with a lower power

consumption. GaN HEMTs on SiC substrates are promising candidates for solving these issues of HPA because of their high-power capabilities and high-efficiency characteristics. To obtain a compact size and high performance, monolithic microwave integrated circuits (MMICs) are becoming more common recently. To date, many X-band GaN-MMIC HPAs with a high output power and efficiency have been reported [75]–[85].

In this study, a high-efficiency X-band GaN-MMIC HPA has been designed and measured. A design technique for an output-matching circuit to obtain high power and efficiency characteristics considering the relation between the optimum load impedance and loss is described. Measured performance shows an output power of 46.1–47.4 dBm (41–56 W), a power added efficiency (PAE) of 49–55% and a gain of 10.1–11.0 dB at 8.5–10.5 GHz, which are obtained with a drain voltage of 30 V. Output power of 47.2–48.4 dBm (53–70 W), PAE of 52–54%, and gain of 11.2–12.1 dB are obtained at the frequency 8.5–10.5 GHz with a drain voltage of 35 V. The developed GaN-MMIC HPA achieves the highest performance in terms of combination of saturated output power and efficiency compared to existing state-of-the-art X-band GaN-MMIC HPAs.

4.2 Individual source via structure

Improvement of transistor structure and optimization of circuit design are important to achieve high efficiency. The ISV structure [86] has been applied to improve the transistor structure, which is described below. Fig. 4-1 shows the schematic of FET structures. Fig. 4-1(a) shows the conventional FET structure. Comb electrode structure with gate fingers between multiple source and drain fingers is typically used for high-power amplifiers. RF signals are distributed from the RF input port to the multiple gate fingers, each of which amplifies the RF signal and outputs it through the drain fingers to the RF output port. Source fingers are bundled and connected to via holes on both sides of the FET. In this

conventional structure, the gain and efficiency characteristics deteriorate at higher frequencies due to parasitic resistance and inductance in the transmission line from the source finger to the via hole. The ISV structure is used to solve this problem. Fig. 4-1(b) shows the FET with ISV structure. The ISV structure has fine via holes in each source finger section, which has the effect of reducing parasitic resistance and inductance of the source and improving gain and efficiency characteristics. Although the structure with smaller vias holes on both sides is effective in reducing inductance, the effect of the ISV structure is dominant because the inductance of the source fingers and drawer lines is more significant. Such structures had been studied earlier in silicon with a structure called TSV structure, but it was difficult to form via holes in small source fingers because of the ISV was made possible by a process using metal masks. In this study, FETs with ISV structure could be fabricated by improving the process, an amplifier design was conducted using them.



(a) Conventional FET structure



(b) FET with ISV structure Fig. 4-1. Schematic of FET structures

4.3 GaN HEMT characteristics

Mitsubishi's 0.15-µm GaN-MMIC is fabricated. The process technology has an AlGaN/GaN HEMT epitaxial layer with an AlN spacer, which improve the electron mobility, grown on a 50-µm-thick SiC substrate. Typical DC characteristics of the FET are $I_{max} = 1.37$ A/mm, gate-drain breakdown voltage exceeding 160 V at $I_{gd} = 1$ mA/mm and three terminal breakdown voltage at pinch-off exceeding 160 V. An individual source via (ISV) structure [86] is employed to reduce source inductance. Fig. 4-2 shows measured load-pull contour of a 10 × 100 µm GaN FET. The measurement is implemented at 10 GHz, a drain voltage of 30 V, a quiescent drain current density of 50 mA/mm, a pulse width of 100 µs and a pulse duty-cycle of 10%. Fig. 4-3 shows measured output power, PAE and gain of the GaN FET. The source and the load impedances are tuned to the PAE maximum impedance in the power sweep measurement and a saturated output power of 37.3 dBm, a PAE of 73%, and an associated gain of 14 dB are obtained. The

improvement due to the ISV structure is estimated to be about 2 dB gain and 2% PAE from the large signal model.



Fig. 4-2. Measured load-pull contour of the GaN FET.



Fig. 4-3. Measured output power, PAE and gain of the GaN FET.

4.4 Design of GaN-MMIC HPA

The design targets of a GaN-MMIC HPA are an output power of 40 W, a gain of over 10 dB, and a PAE of higher than 50% over 8.5-10.5 GHz. To achieve these targets, singlestage amplifiers with eight-unit-cell FETs with a gate width of $10 \times 100 \,\mu\text{m}$ are utilized. The total gate periphery is 8 mm. The design of GaN-MMIC HPA to obtain this wideband high-output-power and high-efficiency characteristics is shown in detail. Fig. 4-4 illustrates a schematic of the GaN-MMIC HPA. Half of the circuit is shown, and the other half is symmetric to the part shown. Fig. 4-5 shows the impedances seen from the part of the output matching circuit of GaN-MMIC HPA indicated in Fig. 4-4 and the optimum load impedances for PAE and the output power for 8.5–10.5 GHz. Each impedance, Z₀, Z₁, Z₂, Z₃ and Z₄, shown as red line in Fig. 4-5 corresponds to the impedance seen from the part of the circuits indicated in Fig. 4-4 for 1-cell FET. The optimum load impedance, PAE maximum impedance, and output power maximum impedance are shown as a blue dashed line and a green dashed line. They are conjugate to an equivalent parallel RC network derived from the load-pull measurement results. The arrows indicate the direction from low frequency to high frequency. The impedance transformation is explained using Fig. 4-4 and Fig. 4-5. Since eight FETs are symmetrically combined, 50 Ω termination (Z₀) seen from 1-cell FET is 400 Ω . Therefore, impedance transformation from 400 Ω to the optimum load impedance is required. From the output port, a DC cut capacitor, a shunt capacitor, transmission lines, shunt inductors, which are also utilized as drain bias circuits and transmission line combiners, are placed, as shown in Fig. 4-4. First, the shunt capacitor and the transmission line transform 400 Ω (Z₀) to a lower real impedance (Z_2) at the center frequency, as shown in Fig. 4-5. This impedance transformation can also be done by a $\lambda/4$ wavelength transmission line but the

combination of a shunt capacitor and a transmission line realizes a comparable transformation in a more compact area. The shunt inductors and series transmission line combiners transform the impedance from Z_2 to Z_4 . By carefully choosing the lengths of the shunt inductors and the series transmission lines, either PAE maximum impedance or output power (P_{out}) maximum impedance can be achieved. Two circuits are designed to obtain either PAE maximum impedance or Pout maximum impedance and their losses are compared. Fig. 4-6 shows the losses of the output matching circuits used to obtain PAE maximum impedance or Pout maximum impedance. The loss of the output matching circuit used to obtain PAE maximum is larger by 0.1 dB than that of the circuit used to obtain Pout maximum. As this increase in loss reduces the merit of the maximum PAE impedance, the circuit with the maximum Pout impedance is chosen as the design target. To compensate for the higher loss at the edge of the operation frequency, the circuit is designed to provide better matching at the edge of the operation frequency. Harmonic impedance is not considered in the design because the dependence of the harmonic impedance was confirmed to be negligibly small in the operation frequency by simulation. The input matching circuit is composed of shunt R-L with DC cut capacitors, which are utilized as gate bias circuits and stabilization circuits, three high-pass and two low-pass impedance transformers. Resistors are placed between each FET to avoid odd-mode oscillation and to realize gate biasing to all FETs. The simulated gate current at saturation is confirmed to be small enough not to affect the RF characteristics. DC-decoupling capacitor is necessary at the input port of the HPA if the output port of the previous device is not DC-decoupled.

Fig. 4-7 shows a photograph of the fabricated X-band GaN-MMIC HPA. The chip area is $3.75 \text{ mm} \times 3.25 \text{ mm}$. A compact design is achieved compared with reported HPAs with a similar output power.



Fig. 4-4. Schematic of the GaN-MMIC HPA. Half of the circuit is shown, and the other half is symmetric to this part.

Impedance seen from the circuit indicated in Fig.2 for 1-cell FET
 PAE maximum impedance for 1-cell FET

Pout maximum impedance for 1-cell FET



Fig. 4-5. Impedance of the output matching circuit for 1-cell FET and simulated optimum load impedance of 1- cell FET for 8.5–10.5 GHz.



Fig. 4-6. Losses of the output matching circuits used to obtain maximum PAE and maximum output power.



Fig. 4-7. Photograph of developed GaN-MMIC HPA.

4.5 Measurement results

Small- and large-signal measurements of the fabricated GaN-MMIC HPA are performed. The measurement conditions are a quiescent drain current density of 50 mA/mm, a pulse width of 100 μ s, and a pulse duty-cycle of 10% at the ambient temperature of 25°C. Fig. 4-8 shows the measured s-parameters of the GaN-MMIC HPA with a drain voltage of 30 V. A linear gain of 12.9–13.8 dB and an input return loss of larger than 5.2 dB over 8.5–10.5 GHz is obtained. Fig. 4-9 shows the measured output power, PAE and gain of the developed GaN-MMIC HPA as a function of input power with a drain voltage of 30 V. The maximum output power of 47.4 dBm and PAE of 55% are obtained at 8.5 GHz. Frequency dependence is measured with a drain voltage of 35 V in addition to 30 V to confirm power capability. Fig. 4-10 shows the measured saturated output power, PAE, and associated gain as a function of frequency. Output power of 46.1– 47.4 dBm (41–56 W), PAE of 49–55%, and gain of 10.1–11.0 dB are obtained at 8.5–10.5 GHz with a drain voltage of 30 V. Output power of 47.2–48.4 dBm (53–70 W), PAE of 52–54%, and gain of 11.2–12.1 dB are obtained at 8.5–10.5 GHz with a drain voltage of 35 V. Broadband high-power and high-efficiency characteristics are obtained. Output power density of 3.3–4.5 W/mm² for a drain voltage of 30 V and 4.3–5.7 W/mm² for 35 V are obtained. Power density over gate periphery is 5.1–7.0 W/mm for a drain voltage of 30 V and 6.6–8.9 W/mm for 35 V. Measured results are compared to other published X-band MMIC HPAs. Table 4-1 shows a comparison of state-of-the-art X-band MMIC HPAs. The measured performances demonstrate the highest performance in terms of the combinations of output power and PAE compared to existing state-of-the-art X-band MMIC HPAs.



Fig. 4-8. Measured s-parameters of GaN-MMIC HPA.



Fig. 4-9. Measured output power, PAE and gain of the developed GaN-MMIC HPA as a function of input power with drain voltage of 30 V.



Fig. 4-10. Measured output power, PAE and gain of the developed GaN-MMIC HPA as a function of frequency. Blue lines are for drain voltage of 30 V and red lines are for drain voltage of 35 V.

Ref	Frequency [GHz]	Output power [W]	PAE [%]	Gain [dB]	Drain Voltage [V]	Area [mm ²]	Output power/Area [W/mm ²]
This	8.5–10.5	41–56	49–55	10.1-11.0	30	12.2	4.5
work	8.5–10.5	53-70	52–54	11.2–12.1	35	12.2	5.7
[75]	10.2	9.2	61.1	10.8	44	NA	NA
[76]	10.0–10.5	10–14	48–61	19	25	9.2	1.52
[77]	9–11	35–45	40–52	16.3	25	18	2.5
[78]	8–12	56–74	40-45	21.5-22.7	28	13.3	5.57

Table 4-1. Comparison of state-of-the-art X-band MMIC HPAs

4.6 Conclusion

An X-band MMIC HPA utilizing the 0.15-µm GaN FET process with an ISV structure is designed and studied. The developed HPA demonstrates an output power of 46.1–47.4 dBm, PAE of 49–55%, and a gain of 10.1–11.0 dB at 8.5–10.5 GHz with a drain voltage of 30 V. Output power of 47.2–48.4 dBm, PAE of 52–54%, and gain of 11.2–12.1 dB are obtained at 8.5–10.5 GHz with a drain voltage of 35 V. The developed GaN-MMIC HPA demonstrates the highest performance in terms of the combination of output power and PAE compared to other reported state-of-the-art X-band MMIC HPAs.

The ISV structure and low-loss output matching circuit technology are more effective at high frequencies where parasitic inductance and dielectric loss are large, and can be expected to be applied to millimeter-wave HPAs. The effect of improvement in efficiency at Ka-band will be between 2-7% [86].

Chapter 5.

Power amplifiers and a switch incorporating partial MMIC and GaN-on-Si technologies for low-cost transmit/receive module

It was shown in Chapters 3 and 4 that GaN-MMICs are compact and offer a high performance thanks to its high degree of freedom in circuit design. However, the cost of GaN-on-SiC substrates tends to be high. Since MMIC chipsets such as power amplifiers and switches account for a high cost ratio of the T/R module, it is important to lower the cost of the chipset. In this chapter, we propose two cost reduction strategies for the T/R modules. One is to use GaN-on-Si devices [88][89]. The other is to use a partial MMIC configuration. Since GaN-on-SiC is expensive, a low-cost and low-loss GaAs MMIC is used as a matching circuit to reduce the cost. Based on these concepts, GaN-on-Si-MMIC DA and HPA, GaN-on-SiC HPA and GaN-on-Si-MMIC SW have been developed as chip sets for low-cost T/R modules.

5.1 Introduction

Transmit-receive (T/R) modules have been widely utilized in applications such as radar technology and communication systems. Active phased array antennas (APAAs), which enable electrical beam formation, have been applied to radar systems [90] and recently also introduced to communication systems such as fifth-generation mobile communications (5G) [91]. As APAAs require thousands of T/R modules, the cost of T/R

modules accounts for a large portion of the system cost [92]. There is an increasing demand for cost-effective transmit modules with high-performance characteristics. A major contributing factor to the cost of T/R modules is the cost of the monolithic microwave integrated circuits (MMICs) component. As the cost of MMIC is mainly driven by the choice of substrate materials and chip size, cost-effective substrate materials and compact design configurations ought to be considered. To realize high power density outputs and PAE, GaN-on-SiC HPAs [93]–[99] and switches (SWs) [100][101] are often utilized. To achieve low-cost chip set, GaN-on-Si-MMIC DA and HPA, GaN-on-SiC HPA and GaN-on-Si-MMIC SW are developed. The concepts of X-band GaN chipsets are proposed, and the design methods and measurement results are shown in Section 5.2, with their performance and cost-effectiveness evaluated against advanced MMIC based chipset devices in Section 5.3.

5.2 Concepts of GaN Chipsets

A schematic of a targeted T/R module is illustrated in Fig. 5-1. The chipsets include amplifier stage and SW. The target performance of amplifier stage is an output power of 20 W and gain of 30 dB to configure proper amplifier chain with a typical Si core-chip with the saturate output power of about 15 dBm. The target performances of SW are isolation of 15 dB with high power input up to 20 W considering isolation of a circulator. Transmit modules with such performances are conventionally composed of a GaAs or GaN-on-SiC MMIC DA, a GaN-on-SiC HPA, and a GaN-on-SiC SW [103]–[109]. Another option for an amplifier stage is utilizing a high-gain GaN-on-SiC MMIC HPA [109] instead of the combination of a DA and an HPA. These configurations demonstrate good performance due to high power capability and high-efficiency characteristic of GaN FETs. However, the configuration tends to be relatively costly due to the high-cost of GaN-on-SiC substrates. GaAs realizes lower cost to area ratios, but the doubling of sizes is necessary to obtain such a high power of 20 W amplifier because of its lower power density. Two types of T/R modules are proposed to configure low-cost and high performance. Fig. 5-1 illustrates schematics of proposed amplifier stages wherein one of the proposed amplifier stage is a combination of GaN-on-Si-MMIC DA and GaN-on-SiC HPA with GaAs output matching circuit (Case-1) as illustrated in Fig. 5-2. The power density of GaN-on-Si is as high as that of GaN-on-SiC while the cost to size ratio is comparable with that of GaAs. GaN-on-SiC HPA with GaAs output matching circuit retains the high performance of GaN-on-SiC FETs while matching circuits are realized by GaAs to reduce cost. The other module is composed of a high-gain GaN-on-Si HPA (Case-2) as illustrated in Fig. 5-3, a DA and an HPA is integrated in a GaN-on-Si chip, and a GaAs output matching circuit is applied. However, the cost of case-2 is lower than that of case-1, whereas, efficiency of case-2 is slightly lower than that of case-1 due to the difference in FET performance. The GaAs output matching circuit is applied to reduce circuit losses compared to those with GaN-on-Si substrate. GaN-on-Si-MMIC SW is proposed to reduce the cost and realizes that while retaining high power capability. Circuit design and measurement results of these four devices are shown in detail as follows.



Fig. 5-1. Schematic of target GaN T/R module with the chipset



Fig. 5-2. Schematic of the proposed amplifier stage case-1.



Fig. 5-3. Schematic of the proposed amplifier stage case-2.

5.3 GaN-on-Si-MMIC DA

5.3.1 Circuit Design

A GaN-on-Si-MMIC DA is designed and fabricated utilizing a 0.25 μ m GaN-on-Si process. Fig. 5-5 illustrates a schematic of GaN-on-Si-MMIC DA. The DA has three FET stages, with gate peripheries of 0.5 mm for the first stage, 0.7 mm for the second stage, and 1.0 mm × 2 cells for the third stage respectively. R//C stabilization circuits with reduced losses at higher frequencies are utilized to compensate for the gain slope of FETs, which results in the DA with broadband flat gain. As illustrated in Fig. 5-4 from the dense integration of matching circuits in limited spaces, the DA is carefully designed considering electromagnetic coupling between each matching circuit. Fig. 5-5 illustrates a photograph of GaN-on-Si-MMIC DA with a chip size of 3.5 mm × 1.1 mm.



Fig. 5-4. Schematic of the GaN-on-Si-MMIC DA



Fig. 5-5. Photograph of the GaN-on-Si-MMIC DA

5.3.2 Measurement

Small and large-signal measurements of the enhanced GaN-on-Si-MMIC DA are performed with a drain voltage of 32 V and a quiescent drain current density of 50 mA/mm. Fig. 5-6 illustrates measured S-parameters of the developed GaN-on-Si-MMIC DA. A linear gain greater than 28 dB is obtained for frequencies conditions of between 8.5–11.0 GHz. Fig. 5-7 illustrates the DA's measured output power, PAE, and gain as a function of frequency. The measurement is performed under pulsed operation with a pulse duty-cycle of 10%, and a saturated output power of 38.1–38.7 dBm, associate gain of
22.3–24.6 dB, and PAE of 32–36% over 9.0–11.0 GHz are obtained. The GaN-on-Si-MMIC DA achieves high power and high gain with an extremely compact size that is less than half the size of typical GaAs MMIC HPA with similar output power[110].



Fig. 5-6. Measured S-parameters of the developed GaN-on-Si-MMIC DA



Fig. 5-7. Measured output power, PAE, and gain of the improved GaN-on-Si-MMIC DA.

5.4 GaN-on-SiC HPA with GaAs input and output matching circuits

5.4.1 Circuit Design

GaN-on-SiC HPA with GaAs input and output matching circuits are designed and fabricated utilizing the 0.25 µm GaN-on-SiC MMIC and GaAs passive MMIC process. A schematic of the GaN-on-SiC HPA with GaAs matching circuits is illustrated in Fig. 5-8. GaAs input and output matching circuits and a GaN-on-SiC FET chip are connected via bonding wires. With HPAs being one of the most power-consuming components in T/R modules, design of the output matching circuit is important to realize high efficiency utilizing harmonic impedance tuning technique [111]-[117]. Impedance matching at the fundamental frequency is mainly based on a quarter-wavelength transmission line and a shunt inductance. In addition, open stubs with an electrical length of $\lambda/4$ at the second harmonic are used to obtain PAE matching at the second harmonic frequency. Fig. 5-9 illustrates the simulated target impedance and impedance of matching circuits seen from the FETs for the fundamental (solid lines) and the second harmonic (dotted lines) frequency. Fig. 5-9(a) illustrates the source-pull contour and the designed impedance of the input matching circuit seen from the FETs for the second harmonic. The impedance for the maximum PAE and the impedance for the minimum PAE are almost equal. Therefore, the designed impedance is slightly mismatched from the maximum PAE impedance so that the impedance is far from the minimum PAE impedance. Fig. 5-9(b) illustrates impedances of the output matching circuit seen from the FETs. PAE contours are plotted by 2 pt. step. Designed impedance of the output matching circuit seen from the FETs for the fundamental (9-11 GHz) and the second harmonic (18-22 GHz) frequencies are shown, and the direction of the arrows indicate the low frequency to the

high frequency. The HPA behaves like inverse class-F amplifier since only the second harmonic impedance is designed to realize open impedance. It is shown that the fundamental and the second harmonic impedances are well matched to the target impedances. On the input side, impedance matching at the fundamental frequency is achieved by one bandpass and one low-pass configuration. Pre-match small shunt capacitors are incorporated into the GaN-on-SiC FET chip to tune the second harmonic impedance. Fig. 5-10 shows a photograph of the developed GaN-on-SiC HPA. The input matching circuit chip sizes, the FET chip, and the output matching circuit are 1.8 mm \times 3.8 mm, 0.8 mm \times 3.8 mm and 2.0 mm \times 3.8 mm, respectively.



Fig. 5-8. Schematic of GaN-on-SiC HPA with GaAs matching circuits. Half of the circuit is shown, and the other half is symmetric.



Fig. 5-9. Simulated targeted and designed impedances of the matching circuits. (a) Impedance of the input matching circuit. (b) Impedance of the output matching circuit.

GaAs input matching circuit GaAs output matching circuit



GaN-on-SiC FET chip

Fig. 5-10. Photograph of GaN-on-SiC HPA with GaAs matching circuits.

5.4.2 Measurement

Small and large-signal measurements of the developed GaN-on-SiC HPA are performed with a drain voltage of 30 V and the quiescent drain current density of 50 mA/mm. Fig. 5-11 illustrates the measured S-parameters of the developed GaN-on-SiC HPA. The linear gain of 8.1–11.5 dB is obtained at frequencies of over 9.0–11.0 GHz. The large-signal measurement is performed under the pulsed condition with a pulse duty-cycle of 10%. Fig. 5-12 illustrates the measured output power, PAE, and gain of the developed GaN-on-SiC HPA as a function of frequency. The drain voltage is 30 V and the quiescent drain current density is 50 mA/mm. A saturated output power of 42.0–44.5 dBm, associate gain of 7.8–10.7 dB, and PAE of 40–56% for 9.0–11.0 GHz is obtained. These measured results show that the lower cost amplifier exhibits comparable characteristics with all reported advanced amplifiers.

By combining GaN-on-Si-MMIC DA and GaN-on-SiC HPA, an output power in excess of 20 W, gain of higher than 30.8 dB, and PAE greater than 37% is estimated for frequencies of over 9.0–11.0 GHz.



Fig. 5-11. Measured S-parameters of GaN-on-SiC HPA



Fig. 5-12. Measured output power, PAE and gain of the developed GaN-on-SiC HPA

5.5 GaN-on-Si HPA with GaAs output matching circuits

5.5.1 Circuit design

The circuit design and measurements of a high-gain GaN-on-Si HPA are shown. A GaAs matching circuit is employed as the output matching circuit of the final stage amplifier to realize lower losses. As illustrated in Fig. 5-3, the high gain four-stage GaN-on-Si HPA amplifier is composed of the GaN-on-Si amplifier chip and GaAs output matching circuit, which are connected via bonding wires. The aforementioned three-stage GaN-on-Si-MMIC DA, an input matching circuit as well as FETs for the final-stage are incorporated into the GaN-on-Si chip. The final stage-gate periphery of the FETs is 1 mm × 8 cells. The DA is placed vertically with respect to the final-stage amplifier to reduce the chip size and connected with HPA via a 50 Ω transmission line. The output matching circuit is the same as the GaN-on-SiC HPA described in Section 2.3 as the target impedances are quite similar. Although the cost to area ratio of the GaAs and GaN-on-Si is comparable, GaAs exhibits lower dielectric losses, which enables high power and high-

efficiency HPA. Fig. 5-13 illustrates a schematic of the input matching circuit of the finalstage GaN-on-Si HPA, which is newly designed for the HPA. Two high-pass and one lowpass section are utilized. Fig. 5-14 illustrates simulated target and designed second harmonic impedances of the input matching circuits of the GaN-on-Si HPA seen from the final-stage FETs. PAE contours of the second harmonic frequency for 10 GHz are plotted by 2 pt. step. The designed impedance for the second harmonic (18–22 GHz) frequencies are also shown and the direction of the arrows indicates the low frequency to the high frequency. The impedance for the maximum PAE and the impedance for the minimum PAE are close each other. Therefore, the designed impedance is mismatched from the maximum PAE impedance so that the impedance is far from the minimum PAE impedance as in case of the GaN-on-SiC HPA. Fig. 5-15 illustrates a photograph of the GaN-on-Si HPA with a GaAs output matching circuit. The total chip size of the GaN-on-Si chip and the GaAs chip are 3.5 mm × 3.8 mm and 2.0 mm × 3.8 mm, respectively.



Fig. 5-13. Schematic of the input matching circuit of GaN-on-Si HPA. Half of the circuit is shown, and the other half is symmetric.



Fig. 5-14. Simulated target and designed second harmonic impedances of the input matching circuits of the GaN-on-Si HPA.



Fig. 5-15. Photograph of the developed GaN-on-Si HPA

5.5.2 Measurement

Small and large-signal measurements are performed, with a drain voltage is 32 V and the quiescent drain current density of 50 mA/mm being recorded. Fig. 5-16 illustrates the measured S-parameters of the developed GaN-on-Si-MMIC HPA. A linear gain of 38.7–

41.2 dB is obtained over frequencies of 9.0–11.0 GHz. Fig. 5-17 illustrates measured output power, gain, and PAE for the improved GaN-on-Si HPA as a function of frequency. The large-signal measurement is operated under a pulse duty-cycle of 10%. And reports a saturated output power in the 41.8–44.8 dBm range, associated gain of 26.9–29.5 dB and PAE of 26–33% for frequencies of 9.0–11.0 GHz. While the efficiency is not as high as that of GaN-on-SiC HPA, these configurations indicate compact and high output power and gain characteristics. The effect of the machining accuracy on the performances is comparable for the two configurations because the deviation of bonding wire length connected to the output matching circuit is dominant on the performance. The configuration of amplifier stage is in a trade-off between the efficiency and the cost.



Fig. 5-16. Measured S-parameters of the GaN-on-Si HPA



Fig. 5-17. Measured output power, PAE and gain of the GaN-on-Si HPA.

5.6 GaN-on-Si-MMIC High Power SW

5.6.1 Circuit design

A cost-effective GaN-on-Si-MMIC SW is designed to protect the receiving devices such as low noise amplifier (LNA) from transmission or reception of unwanted large signals. Fig. 5-18 illustrates schematics of the developed GaN-on-Si-MMIC SW. Fig. 5-18 (a) illustrates the equivalent circuit of the developed GaN-on-Si-MMIC SW. The SW has an input port and a terminal port which is terminated to 50 Ω . The SW has two FETs, namely FET1 and FET2. FET1 is connected in series between the input and the output ports. A transmission line with the length of $\lambda/4$ at the center frequency followed by shunted FET2 is placed between the input and the terminal ports. A spiral inductor is connected in parallel with each FET. Gate electrodes of the FETs are connected to a common DC voltage supply with high resistance. The operation modes of the SW are controlled by a DC voltage applied to the gates of the FETs, wherein the FET shows off-capacitance when biased at -40 V and on-resistance when biased 0 V. The FETs are biased

at 0 V in receive-mode and -40 V in transmit-mode, respectively. Fig. 5-18(b) and (c) illustrate equivalent circuits of the SW in receive-mode and transmit-mode, respectively. In receive-mode, FETs are biased at 0 V so that they behave as low on-resistance. The equivalent circuit of the receive-mode is illustrated in Fig. 5-18(b). FET1 exhibits limited losses and the impedance of the shunt-connected $\lambda/4$ and FET2 exhibits open impedance looking from the input port at the center frequency. Therefore, it has low loss characteristic from the input to the output port. In transmit-mode, isolation between input and output ports must be high to protect receiving devices from high power transmit signal or noise. FETs are biased at -40 V so that they behave as off-capacitance. The equivalent circuit of the transmit-mode is illustrated in Fig. 5-18(c). Each FET and inductance behave like an open circuit, with external signals from input port going to the terminal port. Then, it demonstrates high isolation characteristics between the input and output ports. Design of gate-width of the FETs is explained. Parallel inductors are neglected for simplification. F-matrix between the RF input and output port of the receive-mode is calculated to be

$$F_{\rm RX} = \begin{pmatrix} 1 & 0\\ \frac{1}{Z} \frac{Z + jZ_{\rm L} \tan\theta}{Z_{\rm L} + jZ \tan\theta} & 1 \end{pmatrix} \begin{pmatrix} 1 & R_{\rm on1}\\ 0 & 1 \end{pmatrix}, Z_{\rm L} = Z_0 //R_{\rm on2}$$
(5-1)

where Z_0 is the port impedance, Z and θ are the characteristic impedance and the electrical length of the $\lambda/4$ line, R_{on1} and R_{on2} are the on-resistance of FET1 and FET2, respectively. By assuming that $\theta = 90$ deg., $R_{on} \cong R_{on1}, R_{on2}, Z_L \cong R_{on}, Z \cong Z_0$ and $(R_{on}/Z_0)^2 \cong 0$, Eq. 5-1 is transformed to

$$F_{\rm RX} \cong \begin{pmatrix} 1 & R_{on} \\ R_{on}/Z_0^2 & 1 \end{pmatrix}.$$
 (5-2)

Then the insertion loss in receive-mode (IL_{RX}) is derived by calculating $|S_{21}|$ using Equation (5-2) as

$$IL_{RX} \cong \frac{1}{1 + R_{on}/Z_0} = \frac{1}{1 + R_{on}/W_g Z_0}$$
(5-3)

where R_{on0} (Ω mm) is the on-resistance of the 1 mm-gate-width FET and W_g (mm) is the gate width of the FET. Similarly, F-matrix between the RF input and output port of the transmit-mode is calculated to be

$$F_{\rm TX} = \begin{pmatrix} 1 & 0\\ \frac{1}{Z} \frac{Z + jZ_{\rm L}tan\theta}{Z_{\rm L} + jZtan\theta} & 1 \end{pmatrix} \begin{pmatrix} 1 & 1/j\omega C_{\rm off1}\\ 0 & 1 \end{pmatrix}, Z_{\rm L} = Z_0 / /j\omega C_{\rm off2}$$
(5-4)

Where C_{off1} is the off-capacitance of the FET1. By assuming that $Z_{\text{L}} \cong Z \cong Z_0$, Equation (5-4) is transformed to

$$F_{\rm TX} \cong \begin{pmatrix} 1 & 1/j\omega C_{\rm off1} \\ 1/Z_0 & 1 + 1/j\omega C_{\rm off1} Z_0 \end{pmatrix}.$$
(5-5)

Then the isolation in transmit-mode (ISO_{TX}) is derived by calculating $|S_{21}|$ using Equation (5-5) as

$$ISO_{TX} \cong \frac{1}{\sqrt{2.25 + (1/\omega C_{off1} Z_0)^2}} = \frac{1}{\sqrt{2.25 + (1/\omega C_{off0} W_g Z_0)^2}}$$
(5-6)

Where C_{off0} (F/mm) is the off-capacitance of the 1 mm-gate-width FET. Insertion loss and isolation are in a trade-off relation according to W_g . Fig. 5-19 shows the dependence of insertion loss and isolation on the gate width of FET according to Equation 5-3 and Equation (5-6) ($Z_0 = 50 \Omega$, $R_{\text{on0}} = 3.7 \Omega$ mm, $C_{\text{off0}} = 0.186$ pF/mm and f = 10 GHz). Large W_g of 1 mm is employed for loss reduction and the value of the inductors is set so that the resonation frequency of C_{off} and the inductance is equal to the center frequency to improve isolation.

A photograph of the GaN-on-Si-MMIC SW, with a chip size of $1.05 \text{ mm} \times 1.95 \text{ mm}$ is illustrated in Fig. 5-20. Transmission lines between the input port and the output port are designed to be as short as possible to reduce losses.



Fig. 5-18. Schematic of GaN-on-Si-MMIC SW. (a) Equivalent circuit of developed GaN-on-Si-MMIC SW. (b) Simplified equivalent circuit of SW in receive-mode. (c) Simplified equivalent circuit of SW in transmit-mode.



Fig. 5-19. Dependence of insertion loss and isolation on the total gate width.



Fig. 5-20. Photograph of the developed GaN-on-Si-MMIC SW

5.6.2 Measurements

Fig. 5-21 shows a measurement insertion loss and isolation of the developed GaN-on-Si-MMIC SW. Insertion losses of 1.1–1.3 dB and isolation of 10.1–14.7 dB are obtained at frequencies of 8.0–11.5 GHz. The measured loss is larger than the calculated loss due to the loss of transmission lines is neglected in the calculation. The measured isolation is larger than the calculated isolation since the parallel inductor is neglected in the calculation. Fig. 5-22 illustrates the measured large-signal characteristic of GaN-on-Si-MMIC SW. Deviation of isolation is less than 0.5 dB, and up to 43.1 dBm input power.



Fig. 5-21. Measured small-signal characteristics of the developed GaN-on-Si SW



Fig. 5-22. Measured large-signal characteristic of GaN-on-Si-MMIC SW

5.7 Evaluation of cost-effectiveness

The cost-efficiency of the developed chipsets is evaluated. Table 5-1 summarizes comparisons of size and the estimated relative cost with other published power amplifiers and switches for X-band T/R modules. Applying a partial MMIC configuration increases bonding wire mounting, but in general the cost of wire mounting is negligibly low (<1%) relative to the chip cost. In case-1, the developed GaN-on-Si-MMIC DA, the GaN-on-SiC HPA with the GaAs output matching circuit, and the GaN-on-Si-MMIC SW are utilized. In case-2, the developed GaN-on-Si high-gain HPA and the GaN-on-Si-MMIC SW are utilized. In case-3, conventional T/R modules composed of the GaAs DA[110], the GaN-on-SiC HPA [93] and the GaN-on-SiC SW [100] ever-reported. The cost ratio of GaAs:GaN-on-SiC is typically said to be 1:2 [106][107]. 50% cost reduction of GaN device fabrication is achieved by using larger wafer. Because large Si wafer is commonly available, cost of GaN-on-Si can be half the cost of GaN-on-SiC. The normalized cost is calculated using a chip size and assumed cost ratio of GaN-on-Si:GaAs:GaN-on-SiC = 1:1:2. By employing the chipset shown in case-1, 46% cost reduction from case-3 is

estimated. The cost of case-2 is estimated to be less than half as expensive as case-3. As the output power is comparable between case-1 and case-2, only the efficiency is a trade-off factor with the cost. If power consumption of system specification was not as tight, for example, the system only required short pulse mode operation, case-2 can be the best candidate for the T/R module.

Table 5-1. Comparisons of size and estimated relative cost with other published power amplifiers and a switch for X-band modules.

		DA		HPA		SW		*Cost
		Substrate	Size	Substrate	Size	Substrate	Size	ratio
			(mm ²)		(mm ²)		(mm ²)	(a.u)
This	Case 1	GaN-on-Si	3.9	GaN-on-SiC/GaAs	3.0/14.4	GaN-on-Si	2	0.54
work	Case 2	-	-	GaN-on-Si/GaAs	13.3/7.6	GaN-on-Si	2	0.47
Case 3		GaAs [110]	8.8	GaN-on-SiC [93]	18	GaN-on-SiC [100]	1.9	1

(*) Assuming cost ratio of GaN-on-Si:GaAs:GaN-on-SiC = 1:1:2

5.8 Conclusion

GaN chipsets for cost-effective X-band 20 W T/R modules are fabricated and measured. The chipsets include four devices, a GaN-on-Si-MMIC DA, a GaN-on-SiC MMIC HPA with a GaAs MMIC input and output matching circuits, a high-gain GaN-on-Si HPA with a GaAs output matching circuit, and a GaN-on-Si high power SW. The developed chipsets are estimated to be about half the cost of a conventional chipset while maintaining performances. Utilizing partial MMIC and GaN-on-Si technologies are more advantageous at sub-6 band frequencies because the effects of transistor performance differences, bonding wires will be lower because of low frequency. On the other hand, application to high frequencies, such as millimeter wave, is expected to be affected by transistor performance degradation and narrowing of bandwidth due to the presence of bonding wires.

Chapter 6.

On-chip matching circuits for RF reflectometry using Si quantum dot charge sensor

This chapter reports on extracting the equivalent circuit model parameters of a physically-defined silicon QD at a cryogenic temperature and designing impedance matching circuits to improve the performance of a charge sensor for RF reflectometry. The I–V characteristics and the S-parameters of the quantum dot device are measured around a Coulomb peak at 4.2 K. The measured results are modeled by an RC parallel circuit, and the model parameters for the quantum dot device were obtained. We considered three impedance matching circuits for RF reflectometry of a quantum dot: shunt C-series L-type, shunt L-series C-type, and shunt L-series L-type. We formulated and compared the sensitivity and bandwidth of RF reflectometry for the three types of circuits. The analysis should be useful for selecting the optimal matching circuit and the circuit parameters for given equivalent circuit parameters and working frequency. This procedure is demonstrated for a quantum dot with the characterized model circuit and simulated performance. These results will facilitate fast semiconductor qubit readout in various quantum dot platforms.

6.1 RF reflectometry

Radio-frequency (RF) reflectometry is highly sensitive and useful for spin qubit readout. RF reflectometry using RF-QPC (quantum point contact) [120]–[122], gate sensor [123]–[127] and RF-QD [128]–[134] have been studied. RF-QPC was first proposed and was replaced by highly sensitive RF-QD as process technology matured. Gate sensor reads changes in capacitance, and while it is compact, it is less sensitive than RF-QD. RF-QD is roughly 100 times more sensitive than RF-QPC and roughly ten times more sensitive than gate sensing, making it a promising method for qubit readout [123]. In the case of physically-defined QDs, the number of gate electrodes for a single quantum dot can be reduced compared to gate-defined quantum dots, allowing for integration even using RF-QDs. Multi-functional cryo-CMOS technology is also being developed for future integration [135]–[138]. Cryo-CMOS reduces the number of external interfaces and is used to achieve gate operation and sensing of the quantum dot spin qubit.

A typical QD charge sensor becomes most sensitive at a resistance of $\sim 1 \text{ M}\Omega$. Realizing a good impedance matching of this sensor resistance to the 50 Ω is crucial to achieving high reflectometry sensitivity. Therefore, it is important to design the impedance matching circuit based on the effective circuit model of the QD charge sensor. However, the QD charge sensor is located in a cryogenic environment where precise calibration of the high-frequency transmission properties is challenging, making detailed evaluations of the QD to extract the equivalent circuit parameters elusive. Furthermore, several types of circuits are proposed or utilized for RF reflectometry measurements of a QD charge sensor, which necessitates design methodology and trade-off analysis between the sensitivity and bandwidth for different circuits.

Here, a physically-defined silicon QD [139]–[141] is fabricated and characterized as a testbed to establish the design scheme of a matching circuit for RF reflectometry. The I– V characteristics and S-parameters of the QD are measured to extract the equivalent circuit parameters at the measurement plane located on the bonding pads of the QD chip. The expected sensitivity and bandwidth are analytically obtained for three simple, typical kinds of matching circuits. The result can be used to obtain the optimal circuit parameters given the model parameters for the QD sensor and the target performance. The circuit design formulations are derived to obtain impedance matching between the 50 Ω input port and the QD equivalent circuit represented by an RC parallel circuit. The sensitivity to changes in the resistance or capacitance and the frequency bandwidth is analyzed and compared among the circuits. A selection criterion for the appropriate circuit in terms of sensitivity and bandwidth is given, and its simulated performance for the quantum dot measured in the experiment is presented.

6.2 Device

The device used here is an n-type physically-defined silicon QD fabricated as previously reported [140][141]. Fig. 6-1 shows a scanning electron microscope image of the physically-defined silicon QD device. The device has an n+- polysilicon top gate on top of an insulating layer which induces electrons in the SOI layer. The sensor QD is connected to the source and drain electrodes, and one side gate located nearby is biased to adjust the potential of the sensor QD. The drain and source contacts of the charge sensor QD are connected to ports 1 and 2 of the vector network analyzer (VNA) via RF-through of bias tees, respectively, and to DC measurement systems via DC-through of bias tees. Fig. 6-2 shows a photograph of the silicon QD chip mounted on a PCB. Each pad on the silicon chip is connected to the external circuitry by bonding wires and transmission lines. SMA connectors are mounted on the back side of the PCB and connected to the measurement system via RF cables. The calibration planes of the S-parameter measurement are at the SMA connectors on the PCB, and the S-parameters between the source and drain bond pads are obtained by a deembedding technique from the measurement results of calibrated S-parameters and the simulated S-parameters between the pads and the SMA connectors on the PCB.



Fig. 6-1. Scanning electron micrograph of QD device.



Si QD chip

Fig. 6-2. Picture of the Si QD chip mounted on a PCB.

6.3 Measurements

A detailed equivalent circuit of a QD can be given by using a parasitic capacitance, a tunnel junction (comprising a parallel circuit of tunnel resistance and tunnel capacitance), and a gate-voltage-dependent current source [144]. While this can reproduce various

electrical properties (including I–V characteristics), for RF reflectometry, it can be simplified to a good approximation as a parallel circuit with only a resistor and a capacitor4. Theoretically, both parameters can be evaluated only by measuring the Sparameters. The reflection coefficient should ideally yield both the resistance (R_{QD}) and the capacitance (C_{QD}) of the QD. R_{QD} can be calculated from the reflection coefficient at a low frequency. When the absolute value of the reflection coefficient at low frequencies, such as near DC, is Γ_0 and the port impedance is Z_0 , R_{QD} is calculated as

$$R_{\rm QD} = \frac{1 + \Gamma_0}{1 - \Gamma_0} Z_0. \tag{6-1}$$

The capacitance of the QD (C_{QD}) is calculated as

$$C_{\rm QD} = -\frac{R_{\rm QD}(\Gamma^2 - 2\Gamma\cos\theta + 1) + \sqrt{R_{\rm QD}^2(\Gamma^2 - 2\Gamma\cos\theta + 1)^2 - 16Z_0^2\Gamma^2\sin^2\theta}}{4\Gamma\omega R_{\rm QD}Z_0\sin\theta}.$$
 (6-2)

where Γ is the absolute value of the reflection coefficient, θ is the reflection phase, and ω is the angular frequency. However, R_{QD} is difficult to extract accurately only from the reflection amplitude because it is very high compared to $Z_0 = 50 \Omega$. Therefore, here, we calculate the resistance from the I–V characteristic.

All measurements were performed at a temperature of 4.2 K for a top-gate voltage of 10 V and a drain voltage of 300 μ V. The I–V characteristic is measured using a lock-in amplifier technique at 71 Hz. I–V amplifier and AC voltmeter are used, and the current values are calculated from their conversion factors. Fig. 6-3 shows drain current's side-gate voltage (V_{sg}) dependence (I_{ds}). The current peaks when the side-gate voltage is swept between -2.5 and -4.0 V: a so-called Coulomb peak of the QD. Fig. 6-4 shows the V_{sg} dependence of the resistance of the charge sensor QD (R_{QD}). Fig. 6-5 compares the phase of S₁₁ for V_{sg} = -3.5 and -4.0 V, for which the change in conductance is large. RF reflectometry can read out the change in the resistance and/or the capacitance of a QD due to the variation in the QD potential or the energy dispersion. However, the V_{sg}

dependence of the reflection phase of the QD is small enough to be neglected in this device, meaning that the reflection phase in the present circuit is insensitive to the resistance change. Its frequency dependence can be well fit by the RC model circuit with $R_{\rm QD} = 1.4 \text{ M}\Omega$ and $C_{\rm QD} = 65 \text{ fF}$ using Equation (6-2), suggesting the model's validity in this frequency range. We note that this value of $C_{\rm QD}$ is smaller than the typical parasitic capacitance (on the order of several hundreds of fF) seen in the conventional reflectometry measurements with bond wires or transmission lines of PCBs included.



Fig. 6-3. Drain current dependence on the side-gate voltage.



Fig. 6-4. Resistance of the charge sensor QD as a function of the side-gate voltage.



Fig. 6-5. Measured reflection phase of the S₁₁. The solid red (dotted blue) line shows the result for $V_{sg} = -3.5$ V, (-4.0 V) with smoothing applied. The dashed black line shows the fit result to the parallel RC circuit model for $V_{sg} = -3.5$ V, with the best values ($R_{QD} = 1.4$ M Ω and $C_{QD} = 65$ fF). The fitting of the result for $V_{sg} = -4.0$ V yields a very similar C_{QD} value.

6.4 Circuit design and analysis

The circuit model obtained above should help to devise the matching circuit towards improved sensing performance. In the following, three simple impedance matching circuits were considered, as schematically shown in Fig. 6-6, and a comparison is given in terms of sensitivity and impedance matching bandwidth. In all types (a, b, and c), the matching circuit has two elements, one series and one shunt. The charge sensor QD is modeled by an RC parallel circuit, parameterized by $R_{\rm QD}$ and $C_{\rm QD}$ which take the values of R_0 and C_0 , respectively, under the default bias condition. The values of the components in the matching circuit for a given working (angular) frequency $\omega = \omega_0$ are then determined to match the input impedance of the overall circuits (Z_x with x = a, b, c) to the port impedance (Z_0). The sensitivity indexes to a change in $R_{\rm QD}$ ($S_{\rm R}$) and a change in $C_{\rm QD}$ ($S_{\rm C}$) for circuit x are defined as

$$S_{\mathrm{R}x} = \frac{\partial |Z_x|}{\partial R_{\mathrm{QD}}} \bigg|_{R_{\mathrm{QD}} = R_0, C_{\mathrm{QD}} = C_0, \omega = \omega_0}$$
(6-3)

$$S_{Cx} = \frac{\partial |Z_x|}{\partial C_{QD}} \bigg|_{R_{QD} = R_0, C_{QD} = C_0, \omega = \omega_0}$$
(6-4)

respectively. The larger the absolute value of the index, the higher the sensitivity to the corresponding change. In addition, the frequency derivative of the impedance

$$D_{\omega x} = \frac{\partial |Z_x|}{\partial \omega} \bigg|_{R_{\rm QD} = R_0, \ C_{\rm QD} = C_0, \omega = \omega_0}$$
(6-5)

can be used to evaluate the impedance matching bandwidth at the default bias. The smaller the absolute value of this value, the wider the impedance matching bandwidth. Calculation results for Circuit (a) are shown as

$$C_a = \frac{1}{\omega_0 R_0} \sqrt{\frac{R_0 - Z_0}{Z_0}} - C_0, \tag{6-6}$$

$$L_a = \frac{\sqrt{(R_0 - Z_0)Z_0}}{\omega_0},$$
 (6-7)

$$|S_{\rm Ra}| = \frac{Z_0}{R_0} - 2\left(\frac{Z_0}{R_0}\right)^2,$$
 (6-8)

$$|S_{Ca}| = 2\omega_0 Z_0 \sqrt{(R_0 - Z_0) Z_0},$$
(6-9)

$$|D_{\omega a}| = \frac{2(R_0 - Z_0)Z_0}{\omega_0 R_0},$$
(6-10)

with restrictions of

$$\omega_0 < \frac{1}{C_0 R_0} \sqrt{\frac{R_0 - Z_0}{Z_0}}.$$
(6-11)

Calculation results for Circuit (b) are shown as

$$C_b = \frac{1}{\omega_0 \sqrt{(R_0 - Z_0)Z_0}},$$
(6-12)

$$L_b = \frac{R_0 Z_0}{\omega_0 (\omega_0 C_0 R_0 Z_0 + \sqrt{(R_0 - Z_0) Z_0})},$$
(6-13)

$$|S_{\rm Rb}| = \frac{Z_0}{R_0} - 2\left(\frac{Z_0}{R_0}\right)^2, \qquad (6-14)$$

$$|S_{\rm Cb}| = 2\omega_0 Z_0 \sqrt{(R_0 - Z_0)Z_0},$$
(6-15)

$$|D_{\omega b}| = \frac{2(R_0 - Z_0)Z_0}{\omega_0 R_0} + 4C_0 Z_0 \sqrt{(R_0 - Z_0)Z_0},$$
(6-16)

without any restrictions. Calculation results for Circuit (c) are shown as $R_0 Z_0$

$$L_{c1} = \frac{R_0 Z_0}{\omega_0 \left(\omega_0 C_0 R_0 Z_0 - \sqrt{(R_0 - Z_0) Z_0}\right)} - C_0, \tag{6-17}$$

$$L_{c2} = \frac{\sqrt{(R_0 - Z_0)Z_0}}{\omega_0},\tag{6-18}$$

$$|S_{\rm Rc}| = \frac{Z_0}{R_0} - 2\left(\frac{Z_0}{R_0}\right)^2,$$
 (6-19)

$$|S_{\rm Cc}| = 2\omega_0 Z_0 \sqrt{(R_0 - Z_0)Z_0}, \tag{6-20}$$

$$|D_{\omega c}| = \frac{2(R_0 - Z_0)Z_0}{\omega_0 R_0} - 4C_0 Z_0 \sqrt{(R_0 - Z_0)Z_0},$$
(6-21)

with restrictions of

$$\omega_0 > \frac{1}{C_0 R_0} \sqrt{\frac{R_0 - Z_0}{Z_0}}.$$
(6-22)

The calculated values of the circuit parameters for Circuits (a), (b), (c) are shown by Equations (6-6) and (6-7), (6-12) and (6-13), and (6-17) and (6-18), respectively. Interestingly, the resistance sensitivity indexes, $|S_{Ra}|$, $|S_{Rb}|$, and $|S_{Rc}|$, are the same values for the three circuit types, as given by Equations (6-8), (6-14), and (6-19). These expressions show that higher sensitivity to change in resistance of QD is obtained by lowering R_0 (closer to Z_0). The capacitance sensitivity indexes, $|S_{Ca}|$, $|S_{Cb}|$, and $|S_{Cc}|$, are also the same for the three circuits, as shown by Equations (6-9), (6-15), and (6-20). Based on the calculation, the capacitance sensitivity will be enhanced by increasing R_0 and ω_0 . We noted that the sensitivity was improved for a smaller value of R_0 for resistance readout. In contrast, the opposite is the case for capacitance readout, meaning that the optimal condition depends on the readout mode. The bandwidth indexes, $|D_{\omega a}|$, $|D_{\omega b}|$, and $|D_{\omega c}|$, are shown by Equations (6-10), (6-16), and (6-21). The frequency bandwidths are wider in the order of Circuits (c), (a), (b). Based on the discussion up to this point, it may seem that Circuit (c) is the best among the three, with the widest bandwidth $|D_{\omega x}|$ and identical sensitivities ($|S_{Rx}|$, $|S_{Cx}|$). However, this circuit cannot always be used, because, for a particular combination of the values of R_0 and C_0 , the shunt inductance value given by Equation (6-17) becomes negative (see also expression (6-22)). Circuit (a) becomes realizable in this parameter regime and will be the best choice among the three. The diagram of the best circuit as a function of R_0 and C_0 is given in Fig. 6-7 for several different values of the operation frequency, $\omega_0/(2\pi)$. In the lower left region of the line, Circuit (a) can be realized, and in the upper right region, Circuit (c) can be realized. A matching circuit with only one series inductor can be realized on the lines. The boundary is at 293 MHz for the extracted RC parameters. Once the equivalent circuit parameters are known, the optimal circuit configuration can be determined accordingly.



Fig. 6-6. Schematics of a matching circuit and a QD charge sensor. QD charge sensors are represented as resistance (R_{QD}) and capacitance (C_{QD}) in parallel. (a) Matching circuit with a shunt capacitor (C_a) and a series inductor (L_a). (b) Matching circuit with a shunt inductor (L_b) and a series capacitor (C_b). (c) Matching circuit with a shunt inductor (L_{c1}) and a series inductor (L_{c2})

Based on the analysis above, an optimal impedance matching circuit can be designed for our QD using the extracted equivalent circuit parameters ($R_0 = 1.4 \text{ M}\Omega$ and $C_0 = 65$ fF. The default bias is V_{sg} = -3.5 V). The port impedance of the reflection measurement is assumed to be 50 Ω , the standard for normal measurement systems. The working frequency is assumed to be 2 GHz. We observed that a higher frequency leads to wider matching bandwidth (see (6-10), (6-16) and (6-21)) and compact circuit parameters, but it comes at the cost of the increased effects of parasitic components. Fig. 6-7 indicates that for the given parameters, Circuit (c) should be optimal in terms of the bandwidth with the resistance sensitivity equivalent to that from Circuit (b) while Circuit (a) cannot realize impedance matching. Using Equations (6-17) and (6-18), the circuit parameters are calculated as $L_{c1} = 114$ nH and $L_{c2} = 666$ nH. These values are about 1/10 times smaller than conventional shunt C-shunt L configuration operated at around 100 MHz. The performance of the designed circuit is checked by small-signal simulation. The reflection is calculated for different values of V_{sg} or R_{QD} , which emulates the effect of the electric field change at the sensor QD. Fig. 6-8 shows the frequency dependence of the reflection amplitude of the designed QD charge sensor for two conditions: $R_{\rm QD} = 1.4 \text{ M}\Omega$ and $C_{\rm QD} = 65$ fF corresponding to $V_{\rm sg} = -3.5$ V (the red solid trace) and $R_{\rm QD} = 5.1$ M Ω and $C_{\rm QD} = 65$ fF corresponding to $V_{\rm sg} = -4.0$ V (the blue dashed trace). When $V_{\rm sg}$ is changed from V_{sg} = -3.5 V to -4.0 V, the reflection amplitude at 2 GHz changes from less than -40 dB to about -4.9 dB (owing to design accuracy and measurement accuracy, there is no predominant difference below -40 dB). The bandwidth of less than -10 dB at V_{sg} = -3.5 V is over 0.8 MHz, which is enough for the readout in 1.3 µs to obtain fidelity of 99% considering the dephasing time of 130 µs [28]. The bandwidth in the previous physically-defined QD sensor with a shunt capacitor-series inductor on PCB was 0.5 MHz (estimated readout fidelity of 98.5%). However, it is improved to 0.8 MHz by applying the on-chip shunt inductor-series inductor [145]. The obtained bandwidth is comparable

to the typical reported results with gate-defined QDs [132]. Controlling the tunneling resistance of physically-defined QDs is difficult and tends to be higher than that of gate-defined QDs, resulting in a narrow bandwidth. This problem is solved by proposing a new circuit configuration. Fig. 6-9 shows the $R_{\rm QD}$ dependence of the reflection amplitude at 2 GHz. The reflection amplitude is changed significantly when the value of $R_{\rm QD}$ is varied from $R_{\rm QD} = 1.4 \text{ M}\Omega$, where the circuit is impedance matched to the 50 Ω system.



Fig. 6-7. Optimal circuit configuration when the resistance and the capacitance of the equivalent circuit of a QD are given. The red solid curve, the blue dashed curve, and the green dotted curve show the boundary between the optimal circuit configurations for the frequency of 100 MHz, 500 MHz, and 1 GHz, respectively.



Fig. 6-8. Dependence of the reflection amplitude on the carrier frequency. The red solid line shows the reflection amplitude when $R_{\text{QD}} = 1.4 \text{ M}\Omega$ and $C_{\text{QD}} = 65 \text{ fF}$ and the blue dashed line shows when $R_{\text{QD}} = 5.1 \text{ M}\Omega$ and $C_{\text{QD}} = 65 \text{ fF}$.



Fig. 6-9. Dependence of the reflection amplitude of the designed QD charge sensor on R_{QD} at 2 GHz with $C_{\text{QD}} = 65$ fF.

6.5 Conclusion

In conclusion, a physically-defined silicon QD device was measured. The equivalent circuit parameters of the QD chip at a cryogenic temperature were extracted from the Sparameter measurements in combination with a deembedding technique. Conventionally, shunt capacitor-series inductor circuits have been formed on PCBs in the matching circuit. We have proposed shunt inductor-series capacitor type and shunt inductor-series inductor type circuits. Design methodologies for the three matching circuits were established, with comparisons made regarding the sensing sensitivity and bandwidth. The design and the simulation of the impedance matching circuit with a series inductor and a shunt inductor were performed. The circuit design with smaller inductance than conventional configuration, which enables on-chip matching circuit and the wide bandwidth required for high-speed readout were realized. These results should facilitate the realization of RF reflectometry of semiconductor qubits on various QD platforms. Further miniaturization is possible by increasing the operation frequency using this analysis. However, because the effect of parasitic inductance is large at high frequency, the accuracy of the circuit design using the approximation of the equivalent circuit of QDs as an RC circuit is reduced. The measurement results here indicate that fitting RC circuits up to at least 4 GHz is feasible. For higher frequencies, extracting the equivalent circuit parameters with the closer measurement plane to the drain and the source of the quantum dots is desirable. The size of the developed matching circuit is compact enough for a noisy intermediatescale quantum device (NISQ) with around 10^3 quantum bits but still large for the realization of fault-tolerant quantum computers with around 10⁷ qubits. Reducing the number of matching circuits by time division readout using cryo-CMOS [135] or miniaturizing circuits using active inductors is necessary for further integration.

Chapter 7.

Conclusion

This Ph.D. thesis presents research on semiconductor MMICs for the advanced information and communication network. Chapter 1 explains the necessity of further development of wireless communication technology and quantum network technology in solving social issues. GaN high-power devices and silicon quantum dots devices for qubit are described as key components. It is also mentioned that research on MMIC technologies to improve their performance and make the size compact is important. Chapter 2 describes the importance of broadband, high efficiency, and low-cost GaN-MMIC HPAs, miniaturization, and broadband of silicon quantum dot charge sensors. The current status of these devices and the required performance were discussed in detail. In Chapter 3, we proposed a bandpass NDPA that achieves broadband and high output power. Two GaN-MMICs, single-end GaN-MMIC bandpass NDPA, and two-way combined GaN-MMIC bandpass NDPA were designed and evaluated. Measurement results of the two-way combined GaN bandpass distributed amplifier show output power of 44.3 to 47.9 dBm (27 to 61 W, 40 W on average), PAE of 24 to 43% (31% on average) at an input power of 38 dBm over 2.5 to 10.0 GHz (relative bandwidth of 120%). The measurement results demonstrated the effectiveness of the bandpass distributed amplifier by achieving the world's highest output power and power density among the other reported wideband MMIC HPAs. In Chapter 4, ISV structure with small via holes in all source fingers was applied to improve the performance of GaN amplifiers. We also optimized an output matching circuit by considering the trade-off relationship between load impedance and circuit loss. Measurement results of the developed X-band GaN-MMIC amplifier s 46.147.4 dBm, PAE of 49–55%, and a gain of 10.1–11.0 dB at 8.5–10.5 GHz with a drain voltage of 30 V. Output power of 47.2–48.4 dBm, PAE of 52–54%, and gain of 11.2–12.1 dB were obtained at 8.5–10.5 GHz with a drain voltage of 35 V. The highest performance in terms of output power and PAE combinations was demonstrated compared to existing state-of-the-art X-band MMIC HPAs. In Chapter 5, we studied the configuration of power amplifiers and a switch T/R module that achieves both low cost and high output power. We demonstrated a GaN-on-Si DA, a high gain GaN-on-Si HPA with a GaAs output matching circuit, GaN-on-Si C HPA with a GaAs matching circuit, GaN-on-Si HPA with GaAs output matching circuit and GaN-on-Si SW. Each device achieved comparable performance at about half the cost of conventional amplifiers.

Based on the findings in Chapters 3 to 5, we estimated the performance of a wideband amplifier for the virtualization of RUs. Fig. 7-1 shows the proposed amplifier configuration for the virtualized RU. The bandpass distributed amplifier shown in Chapter 3 shows a 40-W class amplifier with 120% relative bandwidth. Scaling to the lower frequency side with the same relative bandwidth will be possible because of the lower dielectric loss. It is estimated that a 40-W class amplifier can be realized in the 1200-4800 MHz range. Although it does not cover the entire 800–4800 MHz base station frequency in Sub-6, it provides broadband and high-power characteristics. To further broaden the bandwidth, applying stack amplifiers is one possible approach [62]. If the same output power can be achieved with half the gate width, ideally the parasitic capacitance would be halved, and a doubled relative bandwidth of half the 600–4800 MHz could be achieved in a rough estimate. Because there is a trade-off relationship between frequency bandwidth and the amount of mismatch, it may be possible to achieve a bandwidth of 800–4800 MHz with even flatter output power and PAE characteristics. Additional considerations would be necessary to reduce the effect of performance

degradation due to impedance mismatch between common-source FET and common-gate FET. Applying a dual-gate structure [146]–[149] would be one solution to the problem. The backoff PAE also needs to be improved for communication systems. Envelope tracking technique effectively improves efficiency by varying the drain voltage supplied to the amplifier according to the envelope of the transmitted signal, such that the amplifier always operates near the saturation point. Envelope tracking requires a wider bandwidth envelope amplifier as the communication bandwidth becomes wider, and the efficiency of envelope amplifiers becomes lower in the case of wideband communication such as 5G. The wideband amplifier in Chapter 3 has the maximum saturated PAE (43%) in the band, estimated to be improved to 45% by applying the ISV structure described in Chapter 4. Assuming that the envelope tracking amplifier can achieve the same PAE as at saturation output power, the backoff PAE will be about 35% when the efficiency of the envelope tracking power supply of 78% with a switching frequency of 400 MHz is considered [66]. While this value of PAE may be an acceptable range for a communication system, it is still low compared to narrowband amplifiers [65]. In combination with several other technologies, there is potential for improvement to increase efficiency. One approach is to improve the field plate structure of the FET. FET utilizing source field plate structure with reduced parasitic capacitance [150] and drain-connected field plate [151] shows a 3-5% improvement in PAE. The harmonic tuning technique will also be applicable [79], [111]–[117]. This technique can be applied only to the high-frequency side, not to affect the operation frequency. Utilizing a parallel LC circuit, which is low loss at operation frequency and open impedance at the second harmonic at the highfrequency side of the operation frequency will be effective [79]. The technique may improve PAE by 2-6% according to the result of the second harmonic load-pull simulation given in 5.5.1. By utilizing these technologies, achieving PAE of over 50% at saturated output power and 40% backoff PAE will be realizable. For lower cost, a partial

MMIC configuration combining GaN-on-SiC FETs and GaAs matching circuits can be used to reduce cost without sacrificing performance as shown in Chapter 5. Although there are still issues related to frequency bandwidth, efficiency improvement, and linearity, we showed a path toward full virtualization of the base station. Bandwidth, output power, efficiency, and cost (circuit size) are in a trade-off relationship, and there were limitations in adjusting circuit constants based on existing circuit configurations. We have shown that these trade-off relationships can be improved and performance can be enhanced by proposing new circuit configurations, applying new device structures, and appropriately combining different MMICs. These technologies are expected to be applied not only to amplifiers for telecommunications, but also to various applications such as radar and wireless power transmission.



Fig. 7-1. Proposed amplifier configuration for the virtualized RU

In Chapter 6, the design of compact and wideband on-chip matching circuits of quantum dot charge sensors based on RF reflection measurement toward fast readout of a qubit state is implemented. To reduce the size and broaden the bandwidth of the matching circuit, we extracted the equivalent circuit parameters at the silicon chip of the

quantum dots device. We also formulated circuit constants of three matching circuits, the shunt capacitor-series inductor, shunt inductor-series capacitor, and shunt inductor-series inductor. We compared them in terms of sensitivity and frequency bandwidth. By designing a matching circuit with a newly proposed shunt inductor-series inductor type, wide bandwidth enough for fast readout of qubit states within the dephasing time is obtained. In addition, a design with circuit constants that can be on-chip has been realized, showing the prospects for integrating multi-qubits. The size is compact enough for NISQ with around 10^3 quantum bits but still large for realizing of fault-tolerant quantum computers with around 10⁷ qubits. For further integration, reducing the number of matching circuits by time division readout using cryo-CMOS [135], miniaturizing the circuit components using active inductors [138], or 3D packaging technology is necessary. For high-fidelity spin manipulation and readout, not only circuit improvement but also process improvement is necessary. Charge noise is said to be the limiting factor for phase relaxation time. Charge noise is generated by charge entering and leaving defects and impurities in the device, and it is important to reduce these effects. For silicon quantum dots, dangling bonds at the Si/SiO₂ interface [152] and defects at the interface between the insulating layer under the top gate and Si must be reduced. Optimization of process conditions such as cleaning, surface passivation, deposition, and annealing conditions is necessary. In addition, the balance between the top gate and side gate voltages may also reduce the charge noise [153]. Improving the circuits, the device and the operation are important to increase spin manipulation and readout fidelity. Future research on these topics will bring us closer to the realization of quantum computers by reducing the number of qubits required for quantum error correction.

We proposed new circuit configurations and optimized circuits that were impossible with discrete configurations for the requirements of next-generation communication systems and achieved state-of-the-art results. The requirements for RF components are
expected to become more stringent, and MMIC technology will continue to play an increasingly important role in meeting these requirements. Designing MMICs with higher functionality and integration will become increasingly important. Chiplet and heterogeneous integration have also been studied extensively for yield improvement, cost reduction, and performance optimization [154]. Applying these technologies with interconnects smaller than bonding wire to partial MMICs technologies described in Chapter 5 makes it possible to support even higher frequencies and smaller modules. In the future, it will be more important to consider the appropriate partitioning of chips considering function, cost, and yield. It will also be necessary to consider the performance of MMICs and their cooperative operation to optimize their performance when connected to other chips.

References

- C. E. SHANNON, "A Mathematical Theory of Communication," *The Bell System Technical Journal*, Vol. 27, pp. 379–423, 623–656, July, October, 1948.
- [2] R. Baheti and H. Gill: "Cyber-physical systems," *The Impact of Control Technology*, pp.161–166, 2011.
- Y. Yang and K. Hua, "Emerging Technologies for 5G-Enabled Vehicular Networks," in *IEEE Access*, vol. 7, pp. 181117-181141, 2019
- [4] M. Mozaffari, W. Saad, M. Bennis, Y. -H. Nam and M. Debbah, "A Tutorial on UAVs for Wireless Networks: Applications, Challenges, and Open Problems," in *IEEE Communications Surveys & Tutorials*, vol. 21, no. 3, pp. 2334-2360, 2019,
- [5] E. A. Oyekanlu et al., "A Review of Recent Advances in Automated Guided Vehicle Technologies: Integration Challenges and Research Areas for 5G-Based Smart Manufacturing Applications," in *IEEE Access*, vol. 8, pp. 202312-202353, 2020
- [6] N. Islam, B. Ray and F. Pasandideh, "IoT Based Smart Farming: Are the LPWAN Technologies Suitable for Remote Communication?," 2020 IEEE International Conference on Smart Internet of Things (SmartIoT), Beijing, China, 2020, pp. 270-276
- [7] N. Islam, B. Ray and F. Pasandideh, "IoT Based Smart Farming: Are the LPWAN Technologies Suitable for Remote Communication?," 2020 IEEE International Conference on Smart Internet of Things (SmartIoT), Beijing, China, 2020, pp. 270-276
- [8] Mao, Rukun, "Design of Wireless Communication Networks for Cyber-Physical Systems with Application to Smart Grid. " PhD dissertations, University of

Tennessee, 2013.

- [9] Moreira, C., Tiwari, P., Pandey, H. M., Bruza, P. and Wichert, "A. Quantum-like influence diagrams for decision-making. *Neural Networks*. 132, 190–210. (2020).
- [10] Villalba-Diez, J., González-Marcos, A. & Ordieres-Meré, J. "Quantum cyberphysical systems," *Scientific Report*, 12, 7964 (2022).
- [11] N. Moll, P. Barkoutsos, L. S. Bishop, J. M. Chow, A. Cross, D. J. Egger, S. Filipp,
 A. Fuhrer, J. M. Gambetta, M. Ganzhorn, A. Kandala, A. Mezzacapo, P. Müller,
 W. Riess, G. Salis, J. Smolin, I. Tavernelli, and K. Temme, "Quantum optimization using variational algorithms on near-term quantum devices," *Quantum Science and Technology*. 3 (3): 030503.
- [12] D. Tosh, O. Galindo, V. Kreinovich and O. Kosheleva, "Towards Security of Cyber-Physical Systems using Quantum Computing Algorithms," 2020 IEEE 15th International Conference of System of Systems Engineering (SoSE), pp. 313-320, 2020
- [13] S. Wehner, D. Elkouss, R. Hanson, "Quantum internet: A vision for the road ahead," *Science* 362, 6412 (2018)
- [14] X. Liu, H. Jing, S. Zhou, Z. Zhao, X. Cheng and L. Xu, "Key technologies for 5G co-construction and shared base station data automatic configuration," 2021 *IEEE 20th International Conference on Trust, Security and Privacy in Computing and Communications (TrustCom)*, 2021, pp. 1559-1563,
- [15] D. Harutyunyan and R. Riggio, "How to Migrate From Operational LTE/LTE–A Networks to C–RAN With Minimal Investment?," in *IEEE Transactions on Network and Service Management*, vol. 15, no. 4, pp. 1503-1515, Dec. 2018
- [16] J. Ni and J. Hong, "Compact Continuously Tunable Microstrip Low-Pass Filter," in *IEEE Transactions on Microwave Theory and Techniques*, vol. 61, no. 5, pp. 1793-1800, May 2013

- [17] C. F. Campbell and D. C. Dumka, "Wideband high power GaN on SiC SPDT switch MMICs," in *IEEE MTT-S International Microwave Symposium Digest*, 2010, pp. 145–148.
- [18] W. F. Moulder, K. Sertel, and J. L. Volakis, "Superstrate-enhanced ultrawideband tightly coupled array with resistive FSS," *IEEE Trans. Antennas Propag.*, vol. 60, no. 9, pp. 4166–4172, Sep. 2012.
- K. W. Kobayashi, Y. Chen, I. Smorchkova, R. Tsai, M. Wojtowicz and A. Oki, "A 2 Watt, Sub-dB Noise Figure GaN MMIC LNA-PA Amplifier with Multi-octave Bandwidth from 0.2-8 GHz," 2007 *IEEE/MTT-S International Microwave Symposium*, Honolulu, HI, USA, 2007, pp. 619-622
- [20] S. Wang, M. Roger, J. Sarrazin and C. Lelandais-Perrault, "Augmented Iterative Learning Control for Neural-Network-Based Joint Crest Factor Reduction and Digital Predistortion of Power Amplifiers," in *IEEE Transactions on Microwave Theory and Techniques*, vol. 68, no. 11, pp. 4835-4845, Nov. 2020,
- [21] M. El Hassan, M. Crussière, J. -F. Hélard, Y. Nasser and O. Bazzi, "EVM Closed-Form Expression for OFDM Signals With Tone Reservation-Based PAPR Reduction," in *IEEE Transactions on Wireless Communications*, vol. 19, no. 4, pp. 2352-2366, April 2020
- [22] S. Vadgama, M. Hunukumbure, "Trends in Green Wireless Access Networks," 2011 IEEE International Conference on Communications Workshops (ICC), 2011, pp. 1-5,
- [23] F. Granelli, "Green wireless networking: energy efficiency in wireless networks," IEEE ComSoc Distinguished Lecurer Tour, 2014
- [24] D. W. Runton, B. Trabert, J. B. Shealy and R. Vetury, "History of GaN: High-Power RF Gallium Nitride (GaN) from Infancy to Manufacturable Process and Beyond," in *IEEE Microwave Magazine*, vol. 14, no. 3, pp. 82-93, 2013

- [25] Hua Wang, Tzu-Yuan Huang, Naga Sasikanth Mannem, Jeongseok Lee, Edgar Garay, David Munzer, Edward Liu, Yuqi Liu, Bryan Lin, Mohamed Eleraky, Hossein Jalili, Jeongsoo Park, Sensen Li, Fei Wang, Amr S. Ahmed, Christopher Snyder, Sanghoon Lee, Huy Thong Nguyen, and Michael Edward Duffy Smith, "Power Amplifiers Performance Survey 2000-Present," [Online]. Available: <u>https://gems.ece.gatech.edu/PA_survey.html</u>
- [26] P. W. Shor. "Algorithms for quantum computation: Discrete logarithms and factoring," *Proceeding of 35th IEEE FOCS* 124–134, 1994
- [27] A. Barenco, C. H. Bennett, R. Cleve, D. P. DiVincenzo, N. Margolus, T. Sleator,
 P. Shor, J. A. Smolin, and H. Weinfurter. "Elementary gates for quantum computation," *Phys. Rev. A*, 52, 3457–3467, 1995
- [28] F. Arute, K. Arya, R. Babbush, D. Bacon, J. C. Bardin, R. Barends, R. Biswas, S. Boixo, F. G. S. L. Brandao, D. A. Buell, B. Burkett, Y. Chen, Z. Chen, B. Chiaro, R. Collins, W. Courtney, A. Dunsworth, E. Farhi, B. Foxen, A. Fowler, C. Gidney, M. Giustina, R. Graff, K. Guerin, S. Habegger, M. P. Harrigan, M. J. Hartmann, A. Ho, M. Hoffmann, T. Huang, T. S. Humble, S. V. Isakov, E. Jeffrey, Z. Jiang, D. Kafri, K. Kechedzhi, J. Kelly, P. V. Klimov, S. Knysh, A. Korotkov, F. Kostritsa, D. Landhuis, M. Lindmark, E. Lucero, D. Lyakh, S. Mandrà, J. R. McClean, M. McEwen, A. Megrant, X. Mi, K. Michielsen, M. Mohseni, J. Mutus, O. Naaman, M. Neeley, C. Neill, M. Y. Niu, E. Ostby, A. Petukhov, J. C. Platt, C. Quintana, E. G. Rieffel, P. Roushan, N. C. Rubin, D. Sank, K. J. Satzinger, V. Smelyanskiy, K. J. Sung, M. D. Trevithick, A. Vainsencher, B. Villalonga, T. White, Z. J. Yao, P. Yeh, A. Zalcman, H. Neven, and J. M. Martinis. "Quantum supremacy using a programmable superconducting processor," *Nature*, 574 505, 2019.
- [29] N. Friis, O. Marty, C. Maier, C. Hempel, M. Holzäpfel, P. Jurcevic, M. B. Plenio,M. Huber, C. Roos, R. Blatt, and B. Lanyon, "Observation of Entangled States of

a Fully Controlled 20-Qubit System," Phys. Rev. X 8, 021012, 2018

- [30] L. Egan, D. M. Debroy, C. Noel, A. Risinger, D. Zhu, D. Biswas, M. Newman, M. Li, K. R. Brown, M. Cetina, C. Monroe, "Fault-Tolerant Operation of a Quantum Error-Correction Code," arXiv: 2009.11482, 2020
- [31] S. Slussarenko and G. J. Pryde, "Photonic quantum information processing: A concise review," *Appl. Phys. Lett.* 6, 041303, 2019
- [32] A. G. Fowler, M. Mariantoni, J. M. Martinis, and A. N. Cleland, "Surface codes: Towards practical large-scale quantum computation," *Phys. Rev. A*, 86, 032324, 2012
- [33] W. J. Munro, N. L. Piparo, J. Dias, M. Hanks, and K. Nemoto, "Designing tomorrow's quantum internet," AVS Quantum Sci. 4, 020503, 2022
- [34] N. Gisin, G. Ribordy, W. Tittel, and H. Zbinden, "Quantum cryptography," *Rev. Mod.*, vol 74, 2002
- [35] B. M. Maune, M. G. Borselli, B. Huang, T. D. Ladd, P. W. Deelman, K. S. Holabird,
 A. A. Kiselev, I. Alvarado-Rodriguez, R. S. Ross, A. E. Schmitz, M. Sokolich, C.
 A. Watson, M. F. Gyure, and A. T. Hunter. Coherent singlet-triplet oscillations in
 a silicon-based double quantum dot. Nature, 481 344, 2012.
- [36] J. Yoneda, K. Takeda, T. Otsuka, T. Nakajima, M. R. Delbecq, G. Allison, T. Honda, T. Kodera, S. Oda, Y. Hoshi, N. Usami, K. M. Itoh, and S. Tarucha. "A quantum-dot spin qubit with coherence limited by charge noise and fidelity," higher than 99.9%. Nature Nanotechnology, 13 102, 2018.
- [37] R. J. Schoelkopf, P. Wahlgren, A. A. Kozhevnikov, P. Delsing, and D. E. Prober.
 "The radio-frequency single-electron transistor (rf-set): A fast and ultrasensitive electrometer," *Science*, 280 1238, 1998.
- [38] G. Zimmerli, R. L. Kautz, J. M. Martinis, "Noise in the Coulomb blockade electrometer," *Appl. Phys. Lett.* 61, 237 (1992).

- [39] C. Tsironis, R. Meierer, "Load-pull Tuners Are Frequency Selective," *Microwaves* & *RF*, pp.97-100, 2004
- [40] K. B. Niclas, "On Design and Performance of Lossy Match GaAs MESFET Amplifiers," in *IEEE Transactions on Microwave Theory and Techniques*, vol. 30, no. 11, pp. 1900-1907, Nov. 1982
- [41] K. W. Kobayashi, "An 8-Watt 250-3000 MHz Low Noise GaN MMIC Feedback Amplifier with > +50 dBm OIP3," 2011 IEEE Compound Semiconductor Integrated Circuit Symposium (CSICS), Waikoloa, HI, USA, 2011, pp. 1-4
- [42] S. C. Cripps, "RF Power Amplifiers for Wireless Communications." Norwood, MA: Artech House, 1999.
- [43] D. M. Pozar, "Microwave Engineering." Reading, MA: Addison-Wesley, 1990.
- [44] C. F. Campbell, "Gallium nitride power MMICs–Promise and problems," in *Proc. Int. Workshop on Integrated Nonlinear Microwave and Millimetre-Wave Circuits*, 2015.
- [45] Y. Ayasli, R. L. Mozzi, J. L. Vorhaus, L. D. Reynolds, and R. A. Pucel, "A monolithic GaAs 1-13 GHz traveling-wave amplifier," *IEEE Trans. Microw. Theory Tech.*, vol. 30, no. 7, pp. 976–981, July 1982.
- [46] C. Duperrier, M. Campovecchio, L. Roussel, M. Lajugie, and R. Quere, "New design method of uniform and nonuniform distributed power amplifiers," IEEE Trans. Microw. Theory Tech., vol. 49, no. 12, pp. 2494–2500, Dec. 2001.
- [47] L. P. Kouwenhoven, D. G. Austing, and S. Tarucha. "Few-electron quantum dots." *Rep. Prog. Phys.*, 64, 701 (2001).
- [48] K. Ono, D. G. Austing, Y. Tokura, and S. Tarucha. "Current rectification by pauli exclusion in a weakly coupled double quantum dot system." *Science*, 297
- [49] J. M. Elzerman, R. Hanson, L. H. Willems van Beveren, B. Witkamp, L. M. K.Vandersypen, and L. P. Kouwenhoven. "Single-shot read-out of an individual

electron spin in a quantum dot." Nature, 430 431, 2004.

- [50] A. Noiri, K. Takeda, J. Yoneda, T. Nakajima, T. Kodera, and S. Tarucha, "Radio-Frequency-Detected Fast Charge Sensing in Undoped Silicon Quantum Dots," *Nano Letters*, 20 947 (2020)
- [51] D. Keith, M. G. House, M. B. Donnelly, T. F. Watson, B. Weber, and M. Y. Simmons, "Single-Shot Spin Readout in Semiconductors Near the Shot-Noise Sensitivity Limit," *Physical Review X*, 9 041003 (2019)
- [52] E. J. Connors, J. Nelson, and J. M. Nichol, "Rapid High-Fidelity Spin-State Readout in Si/Si-Ge Quantum Dots via rf Reflectometry," *Physical Review Applied*, 13 024019 (2020)
- [53] J. Yoneda, K. Takeda, A. Noiri, T. Nakajima, S. Li, J. Kamioka, T. Kodera, and S. Tarucha, "Quantum non-demolition readout of an electron spin in silicon," *Nature Communication*, 11 1144 (2020)
- [54] T. Struck, A. Hollmann, F. Schauer, O. Fedorets, A. Schmidbauer, K. Sawano, H. Riemann, N. V. Abrosimov, Ł. Cywiński, D. Bougeard, and L. R. Schreiber, "Low-frequency spin qubit energy splitting noise in highly purified ²⁸Si/SiGe.", *npj Quantum Information*, 6(1):40, December 2020.
- [55] C. Campbell, C. Lee, V. Williams, M.-Y. Kao, H.-Q. Tserng, P. Saunier, and T. Balisteri, "A wideband power amplifier MMIC utilizing GaN on SiC HEMT technology," IEEE J. Solid-State Circuits, vol. 44, no. 10, pp. 2640–2647, Oct. 2009.
- [56] J. J. Komiak, K. Chu, and P. C. Chao, "Decade bandwidth 2 to 20 GHz GaN HEMT power amplifier MMICs in DFP and no FP technology," in IEEE MTT-S Int. Microw. Symp. Dig., pp. 1–4, Jun. 2011.
- [57] P. Tran, M. Smith, L. Callejo, M. Wojtowicz, and M. Siddiqui, "2 to 18 GHz highpower and high-efficiency amplifiers," in IEEE MTT-S Int. Microw. Symp. Dig.,

pp. 1–4, Jun. 2017.

- [58] H. Wu, Q. Lin, L. Zhu, S. Chen, Y. Chen, and L. Hu, "A 2 to 18 GHz Compact High-Gain and High-Power GaN Amplifier," in IEEE MTT-S Int. Microw. Symp. Dig., pp. 1–4, Jun. 2019.
- [59] C. F. Campbell, M. D. Roberg, J. Fain, and S. Nayak, "A 1–8GHz gallium nitride distributed power amplifier MMIC utilizing a Trifilar transformer," in Proc. Europ. Microw. Integr. Circuits Conf., pp. 217–220, Oct. 2016.
- [60] J. J. Komiak, "Design and performance of GaN HEMT power amplifiers with up to decade bandwidth for aerospace and defense applications," presented in IEEE MTT-S Int. Microw. Symp., Jun. 2013.
- [61] J. Kamioka, M. Hangai, R. Komaru, T. Morimoto, Y. Kamo, S. Shintaro "Over 20 W 2.5 to 10.5 GHz wideband two-stage GaN MMIC power amplifier with distributed and reactively-matched amplifier stages," in Proc. Europ. Microw. Integr. Circuits Conf., pp.400-403, Oct. 2019.
- [62] H. Wu, Q. Lin, L. Zhu, S. Chen, Y. Chen and L. Hu, "A 2 to 18 GHz Compact High-Gain and High-Power GaN Amplifier," 2019 IEEE MTT-S International Microwave Symposium (IMS), Boston, MA, USA, 2019, pp. 710-713,
- [63] M. Roberg, M. Pilla, T. R. M. Kywe, R. Flynt, N. Chu, H. Rashtian and O. Momeni,
 "A 20W 2-20 GHz GaN MMIC Power Amplifier Using a Decade Bandwidth Transformer-Based Power Combiner," IEEE Trans. Microw. Theory Tech. vol. 67, no. 3, pp. 1039–1049, Mar. 2019.
- [64] E. Kuwata, A. Sugimoto, C. M. Andersson, S. Sakata, H. Koyama, Y. Kamo, K. Yamanaka, and H. Fukumoto, "C-Ku band ultra broadband GaN MMIC amplifier with 20 W output power," in Proc. Asia–Pacific Microw. Conf., pp. 1558–1561, Dec. 2011.
- [65] S. Sakata, K. Kato, E. Teranishi, T. Sugitani, R. Ma, K. Chuang, Y.-C. Wu, K.

Fukunaga, Y. Komatsuzaki, K. Horiguchi, K. Yamanaka, and S. Shinjo, "A Fully-Integrated GaN Doherty Power Amplifier Module with a Compact Frequency-Dependent Compensation Circuit for 5G massive MIMO Base Stations," 2020 IEEE/MTT-S International Microwave Symposium (IMS), p.712, 2020.

- [66] K. Saiki, S. Sakata, Y. Komatsuzaki and M. Tsuru, "High Speed and High Efficiency GaN Envelope Amplifier with Source-Floating Half-Bridge Switch," 2022 IEEE Topical Conference on RF/Microwave Power Amplifiers for Radio and Wireless Applications (PAWR), pp. 22-25, 2022
- [67] J. Kim, H. Park, S. Lee, J. Kim, W. Lee, C. Lee, and Y. Kwon, "6–18 GHz, 26 W GaN HEMT compact power-combined non-uniform distributed amplifier," Electron. Lett., vol. 52, no. 25, pp. 2040–2042, Dec. 2016.
- [68] P. Dennler, S. Maroldt, R. Quay, and O. Ambacher, "Monolithic three-stage 6–18 GHz high power amplifier with distributed interstage in GaN technology," in Europ. Microw. Integr. Circuits Conf., pp. 29–32, Sep. 2015.
- [69] P. Dennler, R. Quay, and O. Ambacher, "Novel semi-reactively-matched multistage broadband power amplifier architecture for monolithic ICs in GaN technology," in IEEE MTT-S Int. Microw. Symp. Dig., pp. 1–4, Jun. 2013.
- [70] Y. Ayasli, R. L. Mozzi, J. L. Vorhaus, L. D. Reynolds, and R. A. Pucel, "A monolithic GaAs 1-13 GHz traveling-wave amplifier," IEEE Trans. Microw. Theory Tech., vol. 30, no. 7, pp. 976–981, Jul. 1982.
- [71] J. B. Beyer, S. N. Prasad, R. C. Becker, J. E. Norman, and G. K. Hohenwarter,
 "MESFET distributed amplifier design guidelines," IEEE Trans. Microw. Theory
 Tech., vol. 32, no. 3, pp. 268–275, Mar. 1984.
- [72] C. Duperrier, M. Campovecchio, L. Roussel, M. Lajugie, and R. Quéré, "New Design Method of Uniform and Nonuniform Distributed Power Amplifiers," IEEE Trans. on Microw. Tech. vol. 49, No. 12, pp. 2494–2500, Dec. 2001.

- [73] H. W. Bode, Network Analysis and Feedback Amplifier Design, New York: Van Nostrand, 1945.
- [74] M. Motoyoshi, "Through-Silicon Via (TSV)," in *Proceedings of the IEEE*, vol. 97, no. 1, pp. 43-48, Jan. 2009.
- [75] J. Kang and J. -S. Moon, "Highly efficient wideband X-band MMIC class-F power amplifier with cascode FP GaN HEMT," in *Electronics Letters*, vol. 53, no. 17, pp. 1207-1209, 17 8 2017, doi: 10.1049/el.2017.1672.
- [76] D. Sardin, T. Reveyrand, and Z. Popović, "X-band 10 W MMIC high-gain power amplifier with up to 60% PAE," 2014 44th European Microwave Conference, Rome, 2014, pp. 1337-1340, doi: 10.1109/EuMC.2014.6986691.
- S. Piotrowic, Z. Ouarch, E. Chartier, R. Aubry, G. Callet, D. Floriot, J. C. Jacquet,
 O. Jardel, E. Morvan, T. Reveyrand, N. Sarazin, and S. L. Delage, "43 W, 52%
 PAE X-Band AlGaN/GaN HEMTs MMIC amplifiers," 2010 IEEE MTT-S International Microwave Symposium, Anaheim, CA, 2010, pp. 505-508, 2010
- [78] H. –Q. Tao, W. Hong, B. Zhang, and X. Yu, "A Compact 60 W X-Band GaN HEMT Power Amplifier MMIC," in *IEEE Microwave and Wireless Components Letters*, vol. 27, no. 1, pp. 73-75, Jan. 2017, doi: 10.1109/LMWC.2016.2630926.
- [79] J. Kamioka, E. Kuwata, M. Hangai, K. Nakahara, Y. Kamo, S. Shinjo, M. Shimozawa, "A 26-W X-Band High Efficiency GaN MMIC Power Amplifier with Compact Spurious Suppression Filters," 2018 Asia-Pacific Microwave Conference (APMC), Kyoto, 2018, pp. 1211-1213
- [80] N. Ono, T. Senju, and K. Takagi, "53% PAE 32 W Miniaturized X-Band GaN HEMT Power Amplifier MMICs," 2018 Asia-Pacific Microwave Conference (APMC), Kyoto, 2018, pp. 557-559
- [81] S. Masuda, M. Yamada, Y. Kamada, T. Ohki, K. Makiyama, N. Okamoto, K. Imanishi, T. Kikkawa, and H. Shigematsu, "GaN single-chip transceiver frontend

MMIC for X-band applications," 2012 IEEE/MTT-S International Microwave Symposium Digest, Montreal, QC, 2012, pp. 1-3

- [82] A. M. Couturier, N. Poitrenaud, V. Serru, R. Dionisio, J. J. Fontecavc, and M. Camiade, "50 % High Efficiency X-Band GaN MMIC Amplifier for Space Applications," 2018 48th European Microwave Conference (EuMC), Madrid, 2018, pp. 352-355
- [83] A. Gurdal, U. Ozipek, B. Sutbas, and E. Ozbay, "A High Gain and High Efficiency
 15 W X-Band GaN Power Amplifier MMIC," 2019 European Microwave
 Conference in Central Europe (EuMCE), Prague, Czech Republic, 2019, pp. 10 13.
- [84] D. Shin, I. Yom, and D. Kim, "X-band GaN MMIC power amplifier for the SSPA of a SAR system," 2017 IEEE International Symposium on Radio-Frequency Integration Technology (RFIT), Seoul, 2017, pp. 93-95
- [85] S. Piotrowicz, E. Morvan, R. Aubry, S. Bansropun, T. Bouvet, E. Chartier, T. Dean, O. Drisse, C. Dua, D. Floriot, M. A. diForte-Poisson, Y. Gourdel, A. J. Hydes, J. C. Jacquet, O. Jardel, D. Lancereau, J. O. Mc Lean, G. Lecoustre, A. Martin, Z. Ouarch, T. Reveyrand, M. Richard, N. Sarazin, D. Thenot, and S. L. Delage, "State of the Art 58W, 38% PAE X-Band AlGaN/GaN HEMTs Microstrip MMIC Amplifiers," 2008 IEEE Compound Semiconductor Integrated Circuits Symposium, Monterey, CA, 2008, pp. 1-4
- [86] D. Tsunami, E. Yagyu, Y. Kamo, K. Nishiguchi, K. Miki, "High-efficiency Kaband GaN High Electron Mobility Transistro for Space Applications," *Mitsubishi Electric technical report*, vol 95, No.6, 2021
- [87] M. Mußer, F. van Raay, P. Brückner, W. Bronner, R. Quay, M. Mikulla, O. Ambacher et al., "Individual source vias for GaN HEMT power bars," 2013 European Microwave Integrated Circuit Conference, Nuremberg, 2013, pp. 184-

187.

- [88] M. -S. Lee, D. Kim, S. Eom, H. -Y. Cha and K. -S. Seo, "A Compact 30-W AlGaN/GaN HEMTs on Silicon Substrate With Output Power Density of 8.1 W/mm at 8 GHz," in *IEEE Electron Device Letters*, vol. 35, no. 10, pp. 995-997, Oct. 2014
- [89] J. -C. Gerbedoen et al., "AlGaN/GaN HEMTs on (001) Silicon Substrate With Power Density Performance of 2.9 W/mm at 10 GHz," in *IEEE Transactions on Electron Devices*, vol. 57, no. 7, pp. 1497-1503, July 2010
- [90] G. W. Stimson, "Introduction to Airborne Radar," 2nd ed., Mendham, NJ, USA: SciTech, 1998, pp.473–491
- [91] S. Shinjo, K. Nakatani, J. Kamioka, R. Komaru, H. Noto, H. Nakamizo, S. Yamaguchi, S. Uchida, A. Okazaki, K. Yamanaka, "A 28GHz-band highly integrated GaAs RF frontend Module for Massive MIMO in 5G," 2018 IEEE MTT-S International Microwave Workshop Series on 5G Hardware and System Technologies (IMWS-5G), Dublin, 2018, pp. 1-3
- [92] J. S. Herd and M. D. Conway, "The Evolution to Modern Phased Array Architectures," in Proceedings of the IEEE, vol. 104, no. 3, pp. 519-529, March 2016
- S. Piotrowicz, Z. Ouarch, E. Chartier, R. Aubry, G. Callet, D. Floriot, J. C. Jacquet,
 O. Jardel, E. Morvan, T. Reveyrand, N. Sarazin, and S.L. Delage "43W, 52%
 PAE X-Band AlGaN/GaN HEMTs MMIC amplifiers," 2010 IEEE MTT-S
 International Microwave Symposium, Anaheim, CA, 2010, pp. 505-508
- [94] T. Torii, S. Imai, H. Maehara, M. Miyashita, T. Kunii, T. Morimoto, A. Inoue, A. Ohta, H. Katayama, N. Yunoue, K. Yamanaka, and H. Fukumoto, "60% PAE, 30W X-band and 33% PAE, 100W Ku-band PAs utilizing 0.15 μm GaN HEMT technology," 2016 46th European Microwave Conference (EuMC), London, 2016,

pp. 568-571

- [95] D. Sardin, T. Reveyrand, and Z. Popović, "X-band 10W MMIC high-gain power amplifier with up to 60% PAE," 2014 44th European Microwave Conference, Rome, 2014, pp. 1337-1340
- [96] J. Kang and J. -S. Moon, "Highly efficient wideband X-band MMIC class-F power amplifier with cascode FP GaN HEMT," in Electronics Letters, vol. 53, no. 17, pp. 1207-1209, 17 8 2017
- [97] H. –Q. Tao, W. Hong, B. Zhang, and X. Yu, "A Compact 60 W X-Band GaN HEMT Power Amplifier MMIC," in IEEE Microwave and Wireless Components Letters, vol. 27, no. 1, pp. 73-75, Jan. 2017
- [98] J. Kamioka, E. Kuwata, M. Hangai, K. Nakahara, Y. Kamo, S. Shinjo, M. Shimozawa, "A 26-W X-Band High Efficiency GaN MMIC Power Amplifier with Compact Spurious Suppression Filters," 2018 Asia-Pacific Microwave Conference (APMC), Kyoto, 2018, pp. 1211-1213
- [99] N. Ono, T. Senju, and K. Takagi, "53% PAE 32 W Miniaturized X-Band GaN HEMT Power Amplifier MMICs," 2018 Asia-Pacific Microwave Conference (APMC), Kyoto, 2018, pp. 557-559
- [100] C. F. Campbell and D. C. Dumka, "Wideband high power GaN on SiC SPDT switch MMICs," 2010 IEEE MTT-S International Microwave Symposium, Anaheim, CA, 2010, pp. 145-148
- [101] M. Hangai, R. Komaru, K. Nakahara, Y. Kamo, M. Hieda and K. Yamanaka, "A high power and low loss GaN HEMT MMIC T/R switch utilizing bandpass/lowpass configuration," 2014 44th European Microwave Conference, Rome, 2014, pp. 1122-1125
- [102] J. Kamioka, Y. Kawamura, Y. Tarui, K. Nakahara, Y. Kamo, H. Okazaki, M. Hangai, K. Yamanaka, and H. Fukumoto, "A Low-Cost 30-W Class X-Band GaN-

on-Si MMIC Power Amplifier with a GaAs MMIC Output Matching Circuit," 2018 13th European Microwave Integrated Circuits Conference (EuMIC), Madrid, 2018, pp. 93-96

- [103] P. Schuh, H. Sledxik, R. Reber, A. Fleckenstein, R. Leberer, M. Oppermann, R. Quay, F. van Raay, M. Seelmann-Eggebert, R. Kiefer, and M. Mikulla, "GaN MMIC based T/R-Module Front-End for X-Band Applications," 2008 European Microwave Integrated Circuit Conference, Amsterdam, 2008, pp. 274-277
- [104] A. Bettidi, D. Carosi, A. Cetronio, F. Corsaro, C. Costrini, C. Lanzieri, and L. Marescialli, "X-band transmit/receive module MMIC chip-set based on emerging GaN and SiGe technologies," IEEE Int. Symp. On phased array systems and technology Dig. Waltham, MA, US, Oct. 2010, pp.250-255
- [105] J. P. B. Janssen, M. van Heijningen, G. Provenzano, G. C. Visser, E. Morvan and F. E. van Vliet, "X-Band Robust AlGaN/GaN Receiver MMICs with over 41 dBm Power Handling," 2008 IEEE Compound Semiconductor Integrated Circuits Symposium, Monterey, CA, 2008, pp. 1-4
- [106] S. Masuda, M. Yamada, T. Ohki, K. Makiyama, N. Okamoto, Y. Nakasha, K. Imanishi, T. Kikkawa, and H. Shigematsu, "C-Ku band GaN MMIC T/R frontend module using multilayer ceramics technology," 2011 IEEE MTT-S International Microwave Symposium, Baltimore, MD, 2011, pp. 1-4
- [107] A. Bettidi, A Cetronio, M. Cicolani, C. Costrini, C. Lanzieri, S. Maccaroni, L. Marescialli, M. Peroni, and P. Romanini, "X-band T/R module in state-of-the-art GaN technology," 2009 European Radar Conference (EuRAD), pp. 258-261, 2009
- [108] S. Masuda, M. Yamada, Y. Kamada, T. Ohki, K. Makiyama, N. Okamoto, K. Imanishi, T. Kikkawa, and H. Shigematsu, "GaN single-chip transceiver frontend MMIC for X-band applications," 2012 IEEE/MTT-S International Microwave Symposium Digest, 2012, pp. 1-3

- [109] S. D'Angelo, A. Biondi, F. Scappaviva, D. Resca and V. A. Monaco, "A GaN MMIC chipset suitable for integration in future X-band spaceborne radar T/R module Frontends," 2016 21st International Conference on Microwave, Radar and Wireless Communications (MIKON), Krakow, 2016, pp. 1-4
- [110] G. Nikandish and A. Medi, "A Design Procedure for High-Efficiency and Compact-Size 5–10-W MMIC Power Amplifiers in GaAs pHEMT Technology," in *IEEE Transactions on Microwave Theory and Techniques*, vol. 61, no. 8, pp. 2922-2933, Aug. 2013
- [111] F. H. Raab, "Class-F power amplifiers with maximally flat waveforms," in *IEEE Transactions on Microwave Theory and Techniques*, vol. 45, no. 11, pp. 2007-2012, Nov. 1997
- [112] R. A. Beltran, "Class-F and inverse class-F power amplifier loading networks design based upon transmission zeros," 2014 IEEE MTT-S International Microwave Symposium (IMS2014), Tampa, FL, 2014, pp. 1-4
- [113] S. Gao, P. Butterworth, S. Ooi and A. Sambell, "High-efficiency power amplifier design including input harmonic termination," in *IEEE Microwave and Wireless Components Letters*, vol. 16, no. 2, pp. 81-83, Feb. 2006
- [114] K. Kuroda, R. Ishikawa and K. Honjo, "Parasitic Compensation Design Technique for a C-Band GaN HEMT Class-F Amplifier," in *IEEE Transactions on Microwave Theory and Techniques*, vol. 58, no. 11, pp. 2741-2750, Nov. 2010
- [115] M. Kamiyama, R. Ishikawa and K. Honjo, "5.65 GHz High-Efficiency GaN HEMT Power Amplifier With Harmonics Treatment up to Fourth Order," in IEEE Microwave and Wireless Components Letters, vol. 22, no. 6, pp. 315-317, June 2012
- [116] S. Tanaka, S. Koizumi, R. Ishikawa, and K. Honjo, "Class-F GaN HEMT Amplifiers Using Compact CRLH Harmonic Tuning Stubs Designed Based on

Negative Order Resonance Modes," in IEICE TRANS. ELECTRON., vol. E102– C, no. 10, pp. 691-698, October 2019

- [117] T. Sugiura, S. Furuta, T. Murakami, K. Tanji, N. Otani, and T. Yoshimasu, "High Efficiency Class-E and Compact Doherty Power Amplifiers with Novel Harmonics Termination for Handset Application," in IEICE TRANS. ELECTRON., vol. E102–C, no. 10, pp. 699-706, October 2019
- [118] W. DeMore, "Gauging the state of GaN power amplification," http://www.analog.com/media/en/technical-documentation/technicalarticles/Gauging-the-State-of-GaN-Power-Amplification.pdf (accessed 2023-02-13)
- [119] R. L. Haupt, "Timed arrays: wideband and time varying antenna arrays," New York: Wiley, 2015, pp.96-98
- [120] D. J. Reilly, C. M. Marcus, M. P. Hanson, and A. C. Gossard, "Fast single-charge sensing with a rf quantum point contact," *Applied Physics Letters*, 91, 162101 (2007)
- K. D. Petersson, C. G. Smith, D. Anderson, P. Atkinson, G. A. C. Jones, and D. A.
 Ritchie, "Charge and Spin State Readout of a Double Quantum Dot Coupled to a Resonator," *Nano Letters*, 10, 2789–2793 (2010)
- [122] T. Müller, B. Küng, S. Hellmüller, P. Studerus, K. Ensslin, T. Ihn, M. Reinwald, and W. Wegscheider, "An in situ tunable radio-frequency quantum point contact," *Applied Physics Letter*. 97, 202104 (2010)
- [123] M. F. Gonzalez-Zalba, S. Barraud, A. J. Ferguson, and A. C. Betz, "[123] the limits of gate-based charge sensing," *Nature Communications*, 6 6084 (2015)
- [124] J. Colless, A. Mahoney, J. Hornibrook, A. Doherty, H. Lu, A. Gossard, and D. Reilly, "Dispersive Readout of a Few-Electron Double Quantum Dot with Fast rf Gate Sensors," *Physical Review Letters*, 110, 046805 (2013).

- [125] A. Crippa, R. Maurand, D. Kotekar-Patil, A. Corna, H. Bohuslavskyi, A. O. Orlov,
 P. Fay, R. Laviéville, S. Barraud, M. Vinet, M. Sanquer, S. D. Franceschi, and X.
 Jehl, "Spectrum and Charge Relaxation in a Silicon Double Quantum Dot Probed
 by Dual-Gate Reflectometry," *Nano Letters*, 17, 1001 (2017).
- [126] J. M. Hornibrook, J. I. Colless, A. C. Mahoney, X. G. Croot, S. Blanvillain, H. Lu,
 A. C. Gossard, and D. J. Reilly, "Frequency multiplexing for readout of spin qubits," *Applied Physics Letters*. 104, 103108 (2014)
- [127] V. N. Ciriano-Tejel, M. A. Fogarty, S. Schaal, L. Hutin, B. Bertrand, L. Ibberson,
 M. F. Gonzalez-Zalba, J. Li, Y.-M. Niquet, M. Vinet, and J. J. L. Morton, "Spin Readout of a CMOS Quantum Dot by Gate Reflectometry and Spin-Dependent Tunneling," *PRX Quantum* 2, 010353 (2021)
- [128] R. J. Schoelkopf, P. Wahlgren, A. A. Kozhevnikov, P. Delsing, and D. E. Prober,
 "The radio-frequency single-electron transistor (RF-SET)," *Science*, 280, 1238 (1998)
- [129] S. J. Angus, A. J. Ferguson, A. S. Dzurak, and R. G. Clark, "A silicon radiofrequency single electron transistor," *Applied Physics Letters*, 92, 112103 (2008)
- [130] M. Jung, M. D. Schroer, K. D. Petersson, and J. R. Petta, "Radio frequency charge sensing in InAs nanowire double quantum dots," *Applied Physics Letters*, 100 253508 (2012)
- [131] N. Ares, F. J. Schupp, A. Mavalankar, G. Rogers, J. Griffiths, G. A. C. Jones, I. Farrer, D. A. Ritchie, C. G. Smith, A. Cottet, G. A. D. Briggs, and E. A. Laird, "Sensitive Radio-Frequency Measurements of a Quantum Dot by Tuning to Perfect Impedance Matching," *Physical Review Applied*, 5, 034011 (2016)
- [132] Y.-Y. Liu, S. G. J. Philips, L. A. Orona, N. Samkharadze, T. McJunkin, E. R. MacQuarrie, M. A. Eriksson, L. M. K. Vandersypen, and A. Yacoby, "Radio-Frequency Reflectometry in Silicon-Based Quantum Dots," *Physical Review*

Applied, 16, 014057 (2021).

- [133] M. House, T. Kobayashi, B. Weber, S. Hile, T. Watson, J. Van Der Heijden, S. Rogge, and M. Simmons," Radio frequency measurements of tunnel couplings and singlet-triplet spin states in Si:P quantum dots," *Nature Communication*, 6, 8848 (2015).
- [134] M. G. House, I. Bartlett, P. Pakkiam, M. Koch, E. Peretz, J. van der Heijden, T. Kobayashi, S. Rogge, and M. Y. Simmons, "High-Sensitivity Charge Detection with a Single-Lead Quantum Dot for Scalable Quantum Computation," *Physical Review Applied* 6, 044016 (2016)
- [135] S. Schaal, A. Rossi, V. N. Ciriano-Tejel, T.-Y. Yang, S. Barraud, J. J. L. Morton, and M. F. Gonzalez-Zalba, "A CMOS dynamic random access architecture for radio-frequency readout of quantum devices," *Nature Electronics* 2, 236–242 (2019)
- [136] F. Sebastiano, J. P.G. Van Dijk, P. A. Thart, B. Patra, J. Van Staveren, X. Xue, C. G. Almudever, G. Scappucci, M. Veldhorst, L. M.K. Vandersypen, A. Vladimirescu, M. Babaie, E. Charbon, "Cryo-CMOS interfaces for large-scale quantum computers," 2020 *IEEE International Electron Devices Meeting (IEDM)*, San Francisco, CA, USA, 2020, pp. 25.2.1-25.2.4
- [137] J. Park, S. Subramanian, L. Lampert, T. Mladenov, I. Klotchkov, D. J. Kurian, E. J.-Hernandez, B. P. Esparza, S. R. Kale, A. Beevi K. T., S. P. Premaratne, T. F. Watson, S. Suzuki, M. Rahman, J. B. Timbadiya, S. Soni, and S. Pellerano, "A Fully Integrated Cryo-CMOS SoC for State Manipulation, Readout, and High-Speed Gate Pulsing of Spin Qubits," in IEEE Journal of Solid-State Circuits, vol. 56, no. 11, pp. 3289-3306, Nov. 2021
- [138] L. Le Guevel, G. Billiot, S. De Franceschi, A. Morel, X. Jehl, A.G.M. Jansen, G.Pillonnet, "Compact gate-based read-out of multiplexed quantum devices with a

cryogenic CMOS active inductor," arXiv:2102.04364, 2021

- [139] K. Horibe, T. Kodera, T. Kambara, K. Uchida, and S. Oda, "Key capacitive parameters for designing single-electron transistor charge sensors," *Journal of Applied Physics*. 111, 093715 (2012)
- [140] R. Mizokuchi, S. Bugu, M. Hirayama, J. Yoneda, and T. Kodera, "Radiofrequency single electron transistors in physically defined silicon quantum dots with a sensitive phase response," *Scientific Report*, 11 5863 (2021)
- [141] S. Bugu, S. Nishiyama, K. Kato, Y. Liu, T. Mori, and T. Kodera, "RF reflectometry for readout of charge transition in a physically defined p-channel MOS silicon quantum dot," *Japanese Journal of Applied Physics*, 60 SBBI07 (2021)
- [142] D. V. Averin, and K. K. Likharev, "Coulomb blockade of single-electron tunneling, and coherent oscillations in small tunnel junctions," *Journal of Low Temperature Physics*, 62, 345–373 (1986)
- [143] S. J. Chorley, J. Wabnig, Z. V. Penfold-Fitch, K. D. Petersson, J. Frake, C. G. Smith, and M. R. Buitelaar," Measuring the Complex Admittance of a Carbon Nanotube Double Quantum Dot," *Physical Review Letters* 108, 036802 (2012)
- [144] Y. Yu, J. H. Oh, S. Hun Son, B. H. Choi, S. Hwang and D. Ahn, "Simulation method of transmission-type Radio-Frequency Single-Electron Transistor (RF-SET) by SPICE," 2006 IEEE Nanotechnology Materials and Devices Conference, pp. 554-555 (2006)
- [145] R. Mizokuchi, S. Nishiyama, K. Kato, Y. Liu, S. Murakami, T. Mori, J. Yoneda, and T. Kodera, "Characterization of sensitivity of a p-channel physically-defined silicon quantum-dot RF charge sensor," in 82nd Conference of Japan Society of Applief Physics, 12-244, 2021
- [146] J. S. Moon, R. Grabar, D. Brown, I. Alvarado-Rodriguez, D. Wong, A. Schmitz,">70% Power-Added-Efficiency Dual-Gate, Cascode GaN HEMTs Without

Harmonic Tuning," in *IEEE Electron Device Letters*, vol. 37, no. 3, pp. 272-275, March 2016,

- [147] P. Dennler, R. Quay, P. Brückner, M. Schlechtweg and O. Ambacher, "Watt-level non-uniform distributed 6–37 GHz power amplifier MMIC with dual-gate driver stage in GaN technology," 2014 IEEE Topical Conference on Power Amplifiers for Wireless and Radio Applications (PAWR), Newport Beach, CA, USA, 2014, pp. 37-39,
- [148] R. Quay et al., "Dual-Gate GaN MMICs for MM-Wave Operation," in IEEE Microwave and Wireless Components Letters, vol. 21, no. 2, pp. 95-97, Feb. 2011,
- [149] S. Hazra, A. A. Shuvo and A. G. Bhuiyan, "Electrical Characteristics of Dual Gate AlGaN/GaN High-Electron Mobility Transistors," 2021 5th International Conference on Electrical Information and Communication Technology (EICT), Khulna, Bangladesh, 2021, pp. 1-4
- [150] K. M. Bothe et al., "Improved X-Band Performance and Reliability of a GaN HEMT With Sunken Source Connected Field Plate Design," in *IEEE Electron Device Letters*, vol. 43, no. 3, pp. 354-357, March 2022
- [151] A. Soni, Ajay and M. Shrivastava, "Novel Drain-Connected Field Plate GaN HEMT Designs for Improved VBD–RON Tradeoff and RF PA Performance," in *IEEE Transactions on Electron Devices*, vol. 67, no. 4, pp. 1718-1725, April 2020,
- [152] N. H. Thoan, K. Keunen, V. V. Afanas'ev, and A. Stesmans, "A Interface state energy distribution and Pbdefects at Si(110)/SiO2 interfaces: Comparison to (111) and (100) silicon orientations," *Journal of Applied Physics*, 109, 013710, 2011
- [153] J. Kamioka, T. Kodera, K. Takeda, T. Obata, S. Tarucha, and S. Oda, "Charge noise analysis of metal oxide semiconductor dual-gate Si/SiGe quantum point contacts", *Journal of Applied Physics*, 115, 203709 (2014)
- [154] J. H. Lau, "Recent Advances and Trends in Advanced Packaging," in IEEE

Transactions on Components, Packaging and Manufacturing Technology, vol. 12, no. 2, pp. 228-252, Feb. 2022

Publications

Scientific articles

- J. Kamioka, T. Kodera, K. Takeda, T. Obata, S. Tarucha, S. Oda, "Charge noise analysis of metal oxide semiconductor dual-gate Si/SiGe quantum point contacts," Journal of Applied Physics, Vol. 115, 203709, May 2014.
- J. Kamioka, Y. Tarui, Y. Kamo, and S. Shintaro,
 "54% PAE, 70-W X-Band GaN MMIC Power Amplifier With Individual Source via Structure,"
 IEEE Microwave and Wireless Components Letters, vol. 30, no. 12, pp. 1149-1152, Dec. 2020
- J. Kamioka, Y. Kawamura, R. Komaru, M. Hangai, Y. Kamo, T. Kodera, and S. Shinjo, "X-Band GaN Chipsets for Cost-Effective 20 W T/R Modules,"
 2022 IEICE Transactions on Electronics, vol. E105-C. (2022)
- 4. J. Kamioka, R. Matsuda, R. Mizokuchi, J. Yoneda, and T. Kodera,
 "Evaluation of a Physically Defined Silicon Quantum Dot for Design of Matching Circuit for RF Reflectometry Charge Sensing" to be published in AIP Advances

Co-authored scientific articles

K. Takeda, T. Obata, Y. Fukuoka, W. M. Akhtar, J. Kamioka, T. Kodera, S. Oda, S. Tarucha.

"Characterization and suppression of low-frequency noise in Si/SiGe quantum point contacts and quantum dots,"

Applied Physics Letters, Vol. 102, pp. 123113, Mar. 2013.

 K. Takeda, J. Kamioka, T. Otsuka, J. Yoneda, T. Nakajima, M. R. Delbecq, G. Allison, T. Kodera, Shunri Oda, S. Tarucha.

"A fault-tolerant addressable spin qubit in a natural silicon quantum dot," Science Advances, vol. 2, No. 8, p. e1600694-1-6, Aug. 2016.

J. Yoneda, K. Takeda, A. Noiri, T. Nakajima, S. Li, J. Kamioka, T. Kodera, and S. Tarucha,

"Quantum non-demolition readout of an electron spin in silicon,"

Nature Communications 11, 1144 (2020)

International conferences

- J. Kamioka, T. Kodera, T. Obata, K. Takeda, W. M. Akhtar, S. Tarucha, S. Oda, "Fabrication of MOS structure gate-defined Si/SiGe quantum dot device," IEEE EDS WIMNACT-37. P-3. 2013
- J. Kamioka, T. Kodera, T. Obata, K. Takeda, W. M. Akhtar, S. Tarucha, S. Oda, "Reduction of Charge Noise in Dual-Gate Si/SiGe Quantum Point Contact Mechanisms,"

2013 International Conference on Solid State Devices and Materials (SSDM), E-2-5

- J. Kamioka, E. Kuwata, M. Hangai, K. Nakahara, Y. Kamo, S. Shinjo, M. Shimozawa, "A 26-W X-Band High Efficiency GaN MMIC Power Amplifier with Compact Spurious Suppression Filters," 2018 Asia-Pacific Microwave Conference (APMC), 2018, pp. 1211-1213
- J. Kamioka, Y. Kawamura, Y. Tarui, K. Nakahara, Y. Kamo, H. Okazaki, M. Hangai, K. Yamanaka, H. Fukumoto,

"A Low-Cost 30-W Class X-Band GaN-on-Si MMIC Power Amplifier with a GaAs

MMIC Output Matching Circuit,"

2018 13th European Microwave Integrated Circuits Conference (EuMIC), 2018, pp. 93-96

- J. Kamioka, M. Hangai, R. Komaru, T. Morimoto, Y. Kamo, S. Shinjo,
 "Over 20 W 2.5 to 10.5 GHz wideband two-stage GaN MMIC power amplifier with distributed and reactively-matched amplifier stages,"
 2019 49th European Microwave Conference (EuMC), 2019, pp. 400-403
- 6. J. Kamioka, M. Hangai, S. Miwa, Y. Kamo, S. Shinjo,
 "2.5 to 10.0 GHz Bandpass Non-Uniform Distributed GaN MMIC HPA,"
 2020 IEEE/MTT-S International Microwave Symposium (IMS), 2020, pp. 265-268
- J. Kamioka, K. Nakatani, Y. Yamaguchi, S. Shinjo,
 "Recent Advances in GaN Power Amplifier Design for Millimeter-Wave Applications"
 2020 International Conference on Solid State Devices and Materials (SSDM), Sep

2020 International Conference on Solid State Devices and Materials (SSDM), Sep 27-30, 2020.